# The Investment Performance of Green versus Non-Green Stocks: A Cross-Sectional Analysis

**Fergal Twomey** 

A dissertation submitted in partial fulfillment of the award of Masters in Business Administration (MBA)

National College of Ireland

Submitted to the National College of Ireland, August 2017

### Abstract

The purpose of the research was to compare the investment performance of green and non-green stocks across four investment styles. Recent literature suggests a lack of consensus among researchers in respect of the performance of green investing, Environmental, Social and Governance (ESG) issues, or Socially Responsible Investing (SRI) relative to more traditional forms of investing. Previous research has focused predominantly on the performance of funds thereby taking into account the skills of fund managers in selecting stocks. Additionally, the existing literature's focus has been on SRI relative to traditional investing or green relative to SRI investing. This research therefore contributes to closing a gap in the literature by investigating the performance of green stocks versus non-green stocks in the market.

Monthly investment returns of green portfolios for four investment styles were compared to their non-green equivalent, over four different time periods. Time weighted returns and Sharpe ratios, to measure risk-adjusted returns, were also computed for further analysis.

The research indicated that there was no significant difference between the monthly returns of green portfolios versus non-green portfolios in the four investments styles over all periods investigated. Time weighted returns showed that green portfolios outperformed non-green portfolios in 14 of the 16 style/periods investigated. The Sharpe ratios showed risk-adjusted returns were consistent with the findings of the time-weighted returns. However, the non-green value/mid-cap portfolio was the only portfolio to have positive returns and Sharpe ratio for the 2008-2009 crisis period.

The importance of the research is to discover performance trends of green and nongreen stocks within different investment styles, irrespective of fund manager skill, which may be informative and helpful for investors when choosing which stocks to invest in. *Keywords:* Investment Performance, environmental, ESG, SRI, green stocks, risk-adjusted returns

## Declaration

#### Submission of Thesis and Dissertation

National College of Ireland

Research Students Declaration Form (Thesis/Author Declaration Form)

Name: Fergal Twomey Student Number: 12120111 Degree for which thesis is submitted: MBA

#### Material submitted for award

(a) I declare that the work has been composed by myself.
(b) I declare that all verbatim extracts contained in the thesis have been distinguished by quotation marks and the sources of information specifically acknowledged.
(c) My thesis will be included in electronic format in the College Institutional Repository TRAP (thesis reports and projects)
(d) *Either* \*I declare that no material contained in the thesis has been used in any other submission for an academic award.
Or \*I declare that the following material contained in the thesis form ed part of a submission for the award of

(State the award and the awardingbody and list the material below)

Signature of research student: \_\_\_\_\_

Date: \_\_\_\_\_

#### Submission of Thesis to Norma Smurfit Library, National College of Ireland

Student name:	_Student number:
School:	Course:
Degree to be awarded:	
Title of Thesis:	

One hard bound copy of your thesis will be lodged in the Norma Smurfit Library and will be available for consultation. The electronic copy will be accessible in TRAP (<u>http://trap.ncirl.ie/</u>), the National College of Ireland's Institutional Repository. In accordance with normal academic library practice all theses lodged in the National College of Ireland Institutional Repository (TRAP) are made available on open access.

I agree to a hard bound copy of my thesis being available for consultation in the library. I also agree to an electronic copy of my thesis being made publicly available on the National College of Ireland's Institutional Repository TRAP.

Signature of Candidate:

For completion by the School: The aforementioned thesis was received by\_\_\_\_\_\_ Date:\_\_\_\_\_\_

This signed form must be appended to all hard bound and electronic copies of your thesis submitted to your school

### Acknowledgements

I would like to thank my supervisor, Colette Darcy for her patience, guidance and support throughout this study. I would also like to thank Jonathan Lambert for his help and guidance with SPSS and data analysis.

I would like to thank my employers KBI Global Investors Ltd. for their support in completing the MBA programme, in particular John Caplice and Ger Solan for their encouragement and flexibility over the last two years. Also thank you to Andros Florides, Massimiliano Tondi, Eoin Fahy and Jean Ryan for their helpful advice and guidance. I would like to thank Joe Kavanagh for the large amount of help and clarification on all things performance measurement related.

I would like to thank my brother Michael Twomey and friend John Given for their generous acceptance of my last-minute proof-reading demands. Also thank you to my parents and family for their encouragement.

I would like to thank the MBA class of 2017 and lecturers of the MBA programme, the wonderful and not so wonderful insights and discussions ensured that the last two years were very interesting!

Finally, I would like to thank my wife Elva, daughter Laragh and sons Cathal and Rossa, for their support and understanding throughout the two years. I'm sure they are as glad as I am that it's all over.

# **Table of Contents**

ABSTRACTI			
DECLARATIONIII			
ACKNOWLEDGEMENTSV			
TABLE OF CONTENTSVI			
LIST OF TABLESIX			
LIST OF FIGURESXII			
LIST OF APPENDICESXIII			
LIST OF ABBREVIATIONSXIV			
1 INTRODUCTION1			
2 LITERATURE REVIEW			
2.1 INTRODUCTION			
2.2 ESG, SRI AND GREEN INVESTING			
2.2.1 SRI			
2.2.2 ESG4			
2.2.3 GREEN INVESTING5			
2.2.4 MSCI ESG RATINGS6			
2.3 PERFORMANCE EVALUATION METHODS7			
2.3.1 FUNDS OR STOCKS			
2.3.2 MEASURING RETURNS			
2.3.3 RISK ADJUSTED RETURNS			
2.3.4 Value or Growth (Price-to-book ratio)12			
2.3.5 Market Cap Size13			
2.4 HOW HAVE SRI AND GREEN STOCKS MEASURED UP?			
2.4.1 LACK OF CONSENSUS			
2.5 CONCLUSION & GAP IN LITERATURE16			
3 RESEARCH QUESTION			
3.1 RESEARCH AIMS & OBJECTIVES18			
3.2 HYPOTHESES			

4	MET	HODOLOGY	. 22
4	4.1 M	METHODOLOGICAL CHOICE	. 22
4	4.2 C	DATA COLLECTION	. 24
	4.2.1	STOCKS OR FUNDS	. 24
	4.2.2	MSCI INDEX	. 24
4	4.3 C	DATA ANALYSIS	. 25
	4.3.1	Portfolio Rebalancing	. 25
	4.3.2	Generate Returns	. 28
	4.3.3		. 30
4	4.4 M	METHODOLOGY SUMMARY	. 31
5	RESE	ARCH FINDINGS	. 32
!	5.1 I	NTRODUCTION	. 32
!	5.2 H	HYPOTHESES TESTING	. 32
	5.2.1	GROWTH / LARGE-CAP 3-YEAR RETURNS	. 33
	5.2.2	GROWTH / LARGE-CAP 5-YEAR RETURNS	. 34
	5.2.3	GROWTH / LARGE-CAP 10-YEAR RETURNS	. 36
	5.2.4	GROWTH / LARGE-CAP 2008-2009	. 38
	5.2.5	VALUE / LARGE-CAP 3-YEAR RETURNS	. 40
	5.2.6	Value / Large-Cap 5-Year Returns	. 42
	5.2.7	VALUE / LARGE-CAP 10-YEAR RETURNS	. 43
	5.2.8	Value / Large-Cap 2008-2009	. 45
	5.2.9	GROWTH / MID-CAP 3-YEAR RETURNS	. 47
	5.2.10	GROWTH / MID-CAP 5-YEAR RETURNS	. 49
	5.2.11	GROWTH / MID-CAP 10-YEAR RETURNS	. 51
	5.2.12	GROWTH / MID-CAP 2008-2009	. 53
	5.2.13	VALUE / MID-CAP 3-YEAR RETURNS	. 55
	5.2.14	VALUE / MID-CAP 5-YEAR RETURNS	. 56
	5.2.15	VALUE / MID-CAP 10-YEAR RETURNS	. 58
	5.2.16	VALUE / MID-CAP 2008-2009	. 60
!	5.3 1	TIME-WEIGHTED RETURNS	. 62

5.3.1	GROWTH / LARGE-CAP	62
5.3.2	VALUE / LARGE-CAP	63
5.3.3	GROWTH / MID-CAP	64
5.3.4	VALUE / MID-CAP	65
5.4 S	HARPE RATIO	66
5.4.1	GROWTH / LARGE-CAP	66
5.4.2	VALUE / LARGE-CAP	67
5.4.3	GROWTH / MID-CAP	68
5.4.4	VALUE / MID-CAP	69
6 DISC	USSION	71
7 CON	CLUSION	74
REFERENC	CES	76
APPENDIC	CES	85
APPEND	IX A: DESCRIPTIVE STATISTICS	85
APPEND	IX B: LETTER OF PERMISSION FROM MSCI	101

## **List of Tables**

Table 4.1: Green and Non-Green Portfolios

Table 4.2: Number of stocks per portfolio per year

**Table 5.1**: Test of normality for Green and Non-Green Growth/Large Cap Portfolios 3-Year

Table 5.2: Groups Statistics - GGL versus NGL 3-Year period

Table 5.3: Independent Samples t-Test – GGL versus NGL 3-Year period

**Table 5.4**: Test of normality for Green and Non-Green Growth/Large-Cap Portfolios 5-Year

Table 5.5: Groups Statistics - GGL versus NGL 5-Year period

Table 5.6: Independent Samples t-test – GGL versus NGL 5-Year period

**Table 5.7**: Test of normality for Green and Non-Green Growth/Large Cap Portfolios 10-Year

Table 5.8: Mean Rank – GGL versus NGL 10-Year period

Table 5.9: Mann Whitney U test – GGL versus NGL 10-Year period

**Table 5.10**: Test of normality for Green and Non-Green Growth/Large Cap Portfolios2008-2009

Table 5.11: Groups Statistics – GGL versus NGL 2008-2009 period

Table 5.12: Independent Samples t-Test – GGL versus NGL 2008-2009 period

**Table 5.13**: Test of normality for Green and Non-Green Value/Large Cap Portfolios 3-Year

Table 5.14: Groups Statistics – GVL versus NVL 3-Year period

Table 5.15: Independent Samples t-Test – GVL versus NVL 3-Year period

**Table 5.16**: Test of normality for Green and Non-Green Value/Large Cap Portfolios 5-Year

Table 5.17: Mean Rank – GVL versus NVL 5-Year period

Table 5.18: Mann-Whitney U-test – GVL versus NVL 5-Year period

**Table 5.19**: Test of normality for Green and Non-Green Value/Large Cap Portfolios 10-Year

Table 5.20: Mean Rank – GVL versus NVL 10-Year period

Table 5.21: Mann Whitney U test – GVL versus NVL 10-Year period

 Table 5.22: Test of normality for Green and Non-Green Value/Large Cap Portfolios

 2008-2009

Table 5.23: Groups Statistics – GVL versus NVL 2008-2009 period

Table 5.24: Independent Samples t-Test – GVL versus NVL 2008-2009 period

**Table 5.25**: Test of normality for Green and Non-Green Growth/Mid Cap Portfolios 3-Year

Table 5.26: Groups Statistics – GGM versus NGM 3-Year period

Table 5.27: Independent Samples t-Test – GGM versus NGM 3-Year period

**Table 5.28**: Test of normality for Green and Non-Green Growth/Mid-Cap Portfolios 5-Year

Table 5.29: Group Statistics – GGM versus NGM 5-Year period

 Table 5.30: Independent Samples t-Test – GGM versus NGM 5-Year period

**Table 5.31**: Test of normality for Green and Non-Green Growth/Mid Cap Portfolios 10-Year

Table 5.32: Mean Rank – GGM versus NGM 10-Year period

Table 5.33: Mann Whitney U test – GGM versus NGM 10-Year period

**Table 5.34**: Test of normality for Green and Non-Green Growth/Mid Cap Portfolios2008-2009

Table 5.35: Groups Statistics – GGM versus NGM 2008-2009 period

Table 5.36: Independent Samples t-Test – GGM versus NGM 2008-2009 period

 Table 5.37: Test of normality for Green and Non-Green Value/Mid Cap Portfolios 3 

 Year

Table 5.38: Groups Statistics – GVM versus NVM 3-Year period

Table 5.39: Independent Samples t-Test – GVM versus NVM 3-Year period

**Table 5.40**: Test of normality for Green and Non-Green Value/Mid-Cap Portfolios 5-Year

Table 5.41: Group Statistics – GVM versus NVM 5-Year period

 Table 5.42: Independent Samples t-Test – GVM versus NVM 5-Year period

**Table 5.43**: Test of normality for Green and Non-Green Value/Mid Cap Portfolios 10-Year

Table 5.44: Mean Rank – GVM versus NVM 10-Year period

Table 5.45: Mann Whitney U Test – GVM versus NVM 10-Year period

**Table 5.46**: Test of normality for Green and Non-Green Value/Mid Cap Portfolios 2008-2009

Table 5.47: Groups Statistics – GVM versus NVM 2008-2009 period

Table 5.48: Independent Samples t-Test - GVM versus NVM 2008-2009 period

Table 9.1: GGL versus NGL 3-Year Descriptive Statistics

**Table 9.2:** GGL versus NGL 5-Year Descriptive Statistics

Table 9.3: GGL versus NGL 10-Year Descriptive Statistics

Table 9.4: GGL versus NGL 2008-2009 Descriptive Statistics

Table 9.5: GVL versus NVL 3-Year Descriptive Statistics

 Table 9.6: GVL versus NVL 5-Year Descriptive Statistics

**Table 9.7:** GVL versus NVL 10-Year Descriptive Statistics

Table 9.8: GVM versus NVM 2008-2009 Descriptive Statistics

Table 9.9: GGM versus NGM 3-Year Descriptive Statistics

Table 9.10: GGM versus NGM 5-Year Descriptive Statistics

Table 9.11: GGM versus NGM 10-Year Descriptive Statistics

Table 9.12: GGM versus NGM 2008-2009 Descriptive Statistics

 Table 9.13: GVM versus NVM 3-Year Descriptive Statistics

Table 9.14: GVM versus NVM 5-Year Descriptive Statistics

Table 9.15: GVM versus NVM 10-Year Descriptive Statistics

Table 9.16: GVM versus NVM 2008-2009 Descriptive Statistics

## **List of Figures**

Figure 2.1: A Fund Style Box

Figure 2.2: Sharpe Ratio

Figure 2.3: Fama-French three factor model

Figure 4.1: Time Weighted Returns

**Figure 5.1**: Geometric Returns: Green versus Non-Green for Growth/Large-Cap portfolios

Figure 5.2: Geometric Returns: Green versus Non-Green for Value/Large-Cap portfolios

**Figure 5.3**: Geometric Returns: Green versus Non-Green for Growth/Mid-Cap portfolios

Figure 5.4: Geometric Returns: Green versus Non-Green for Value/Mid-Cap portfolios

Figure 5.5: Sharpe ratio - Green versus Non-Green for Growth/Large-Cap portfolios

Figure 5.6: Sharpe ratio - Green versus Non-Green for Value/Large-Cap portfolios

Figure 5.7: Sharpe ratio - Green versus Non-Green for Growth/Mid-Cap portfolios

Figure 5.8: Sharpe ratio - Green versus Non-Green for Value/Mid-Cap portfolios

# List of Appendices

Appendix A: Descriptive Statistics

Appendix B: Letter of permission from MSCI

# **List of Abbreviations**

AWCI	All World Country Index
САРМ	Capital Asset Pricing Model
ESG	Environment Social Governance
ETF	Exchange-Traded Fund
GGL	Green Growth Large-Cap
GGM	Green Growth Mid-Cap
GVL	Green Value Large-Cap
GVM	Green Value Mid-Cap
FTSE	Financial Times Stock Exchange
HML	High Minus Low
MPT	Modern Portfolio Theory
MSCI	Morgan Stanley Capital International
NGL	Non-green Growth Large-Cap
NGM	Non-green Growth Mid-Cap
NVL	Non-green Value Large-Cap
NVM	Non-green Value Mid-Cap
Р/В	Price-to-book
S&P	Standard & Poor's
SMB	Small Minus Big
SPSS	Statistical Package for the Social Sciences
SQL	Structured Query Language
SR	Sharpe Ratio
SRI	Socially Responsible Investing

### **1** Introduction

According to the Social Investment Forum (2016), \$8.72 trillion assets under professional management in the US utilise Environmental, Social and Governance (ESG) factors. Globally, the Socially Responsible Investing (SRI) market is estimated to be €21.4 trillion (Kłobukowska, 2017). Increasing awareness of environmental issues and climate change has led to policies, regulations and treaties such as the Kyoto Protocol (Chia, Goldberg, Owyong, Shepard & Stoyanov, 2009).

Przychodzen, Gómez-Bezares, Przychodzen & Larreina (2016) highlight the concerns of stakeholders about ESG issues and their demands that the markets address these issues. Kleiner (1991) pointed out over a quarter of century ago that environmental concerns and corporate interests were not mutually exclusive. More recently, Boulatoff & Boyer (2009) posit that environmentalism has increasingly become a part of society and as a strategy, can help organisations gain a competitive advantage. Incidents such as the Mexican Gulf oil spill in 2010 have brought to the attention of stakeholders the impact ESG factors can have on share prices when BP PLC lost 55% of its share value following the disaster (Hua, 2011). As the most recent SRI niche, green investing is growing at a fast rate (Chang, Nelson, & Witte, 2012), but as a subset of SRI, and therefore considered to be further restricted in stock selection, a key issue is whether or not there is a cost or benefit to holding stocks of green-friendly companies (Puopolo, Teti & Milani, 2015). It is clear therefore, that research on the outcome of green investment strategies would be of interest to investors to determine whether favouring green stocks in their investment portfolios comes at a financial cost.

While most studies have focused on either green investment funds versus non-green investment funds, SRI investing versus more traditional investing or the performance of green investing compared to SRI investing, this study attempts to contribute to closing the gap in the literature by comparing the performance of green stocks versus non-green stocks.

The paper is organised into the following chapters:

- **Chapter 1** introduces the topic, a brief background and rationale for research
- Chapter 2 provides a literature review of relevant research, theory and concepts. Identification of research gap and relevance of literature to this research
- Chapter 3 covers the development of research aim, objectives and hypotheses
- Chapter 4 outlines the methodology used to collect and analyse data
- **Chapter 5** sets out the research findings
- Chapter 6 provides a discussion on the research findings
- **Chapter 7** concludes by covering implications, limitations and suggestions for further research

### 2 Literature Review

#### 2.1 Introduction

A literature review on SRI, ESG and green mutual funds and stocks will be undertaken in an attempt to draw out a definition of green investing. The different methods researchers have employed to measure and evaluate investment performance will then be identified. Next, a review of the different models employed to measure risk-adjusted returns will be conducted in an attempt to provide justification for splitting portfolios by size and style. Following on from that, the main findings in the literature of green fund performance versus more traditional funds will be reviewed. Finally, the main focus of existing literature will be discussed in an effort to identify any gaps of interest that can steer the focus of this study.

#### 2.2 ESG, SRI and Green Investing

The terms ESG, SRI and green investing are often used interchangeably to distinguish from traditional finance but their differences are important (Hay, 2015). Lesser, Lobe & Walkshäusl (2014) highlight that the academic literature often mixes up terminology when discussing green investing and SRI. This section therefore will look at the existing literature to identify the similarities and differences between the three terms in an attempt to draw out a definition of green investing.

#### 2.2.1 SRI

SRI is an ethical investing approach where moral criteria are applied which can sometimes allow the asset manager to sacrifice financial returns for ethics (Hay, 2015). SRI has moved from the fringes towards mainstream investment management

(Marlowe, 2014), with the global SRI market estimated to be €21.4 trillion (Kłobukowska, 2017). Napach (2016) describes SRI as a form of investment where the social or environmental concerns are to the forefront. Napach (2016) views both ESG and SRI as sitting somewhere along a continuum from traditional investing, where the sole aim is returns without care for the social dimension toward philanthropic and impact investing, where financial profit is not the primary goal.

SRI funds began by the exclusion of so-called "sin-stocks" issued by firms producing socially undesirable products such as weapons, tobacco and gambling (Berry & Junkus, 2013; Nofsinger & Varma, 2014). SRI is based largely on exclusion of companies, after analysing for certain criteria typically the bottom third would be considered off limits (Prudhomme, 2015). In an attempt to define stocks as SRI or non-SRI, Filbeck, Holzhauer & Zhao (2014) used socially responsible ratings from KLD Research and analytics, now MSCI ESG Research (Semonova & Hassel, 2015) with cut-off points of top 30% and bottom 30% to classify stocks into suitable SRI or non-SRI portfolios.

As no clear classification of whether a stock is green or non-green exists, this research will take a similar approach to the Filbeck et al. (2014) method of using a top 30%, bottom 30% of stock environmental scores from MSCI ESG ratings, to define stocks as green or non-green.

#### 2.2.2 ESG

ESG investing takes environmental (E), social (S) and governance (G) issues into account when choosing stocks but not necessarily with any moral intent (Hay, 2015), i.e. socalled "bad" stocks may still be chosen and justified in financial terms. It attempts to measure the intangible non-financial performance metrics that contribute to market valuation (Boerner, 2007). ESG is not considered to be equivalent to ethical investing or restricted to environmental issues. Napach (2016) describes ESG as prioritising profit first, with environmental, social and governance issues being a secondary concern.

The last decade has seen ESG become increasingly significant to policy makers and investors (Garcia, Medes-Da-Silva & Orsato, 2017). Companies that disclose their ESG practices were considered to have enhanced their reputation and improved investor confidence (Tarmuji, Maelah & Tarmuji, 2016). Responsible investing and good returns are not mutually exclusive, and how much weight a company puts on ESG issues highlights to investors not only the company's values but may also be an indication of how well a company is managed (Carlsson-Sweeny, 2014). Research by Tarmuji et al. (2016) show evidence that ESG practices have a positive impact on economic performance.

#### 2.2.3 Green Investing

Green investing is the latest niche to come from SRI with fewer studies in comparison to other areas, and as a result there is no formal definition (Chang at al., 2012). Mallett & Michelson (2010) use the terms green and sustainable investing interchangeably and describe green investing as a form of investing that appeals to investors interested in climate change, the environment and a sustainable economy. Sabbaghi (2011) adds social and governance traits to the mix in defining green stocks whereas Climent & Soriano (2011) have a narrower definition, describing green funds as ones that seek environmentally responsible investments. Boulatoff & Boyer (2009) in a study of 310 green companies defined green investing as investing in stocks whose companies have a positive environmental impact. Their study focused on eleven environmental industries: Biofuels, Efficiency, Energy Storage, Fuel Cell, Geothermal, Recycling/Green Chemicals, Renewable Energy, Solar, Transportation, Water and Wind Power.

Chang et al. (2012) regard green investing as a smaller subset of both SRI and ESG which therefore impacts the diversification that can be achieved and so may expose investors

to more risk. However, Mallet & Michelson (2010) argue that from a stock diversification point of view, the overlap between green and SRI is unclear, in that if green investing is to be considered a subset of SRI then it would be less diverse. However, if green investing is considered to include environmentally conscious companies only and does not have to take the other social or ethical considerations of ESG into account, then a green fund actually has access to a larger universe of stocks.

#### 2.2.4 MSCI ESG Ratings

In relation to this research, a challenge of how to classify stocks as green has emerged. In highlighting this challenge one particular study (Mallet & Michelson, 2010) determined that if a fund manager calls the fund green then it is green.

Stock market index provider MSCI conducts research on the companies which are listed on the MSCI All Country World Index (MSCI ACWI) with a view to ranking them versus their global industry peer groups on ESG issues. MSCI ESG Ratings is an industry adjusted score from 0 to 10, where 0 is the worst and 10 is the best rating. Individual rating for each of the three E, S and G pillars, without the industry adjustment, are also provided (MSCI, 2014; MSCI, 2017a). The environmental pillar rating is based on several environmental themes, including Climate Change, Natural Resources, Pollution & Waste and Environmental Opportunities (MSCI, 2017a).

MSCI ESG ratings have been used extensively in academic research to hypothesise on the performance of companies based on their environmental ratings (Semenova & Hassel, 2014). Kim & Statman (2012) use KLD ratings, now MSCI ESG Research (Semonova & Hassel, 2015) to measure the effect of ratings on a company's financial performance. Similarly, Albuquerque, Durnev & Koskinan (2013) use MSCI ESG ratings to study the effect of social and environmental risk factors on a company's financial performance and risk management. In the challenge of categorizing and evaluating green investments, using the MSCI ESG environmental pillar ratings would represent a more objective approach than comparing green and non-green funds. To categorise stocks in their relevant green /non-green bucket therefore, this research will use the environmental pillar score of MSCI ESG Ratings. To decide on a cut-off, point for green and non-green, this research will adopt the same top 30% / bottom 30% method that Filbeck et al. (2014) used to categorise SRI and non-SRI investments. A green stock will be classified as all those stocks that are within the top 30% MSCI ESG environmental pillar ratings. A non-green stock will be classified as all those stocks that are within the top 30% MSCI ESG environmental pillar ratings.

#### 2.3 Performance Evaluation Methods

The following section will discuss the methods of data collection and performance measurement as evidenced in the academic literature. Where possible this research will focus mainly on socially responsible investing in general and green investing in particular. However, the methods employed in any research that evaluate and compare the performance of portfolios or funds is of relevance. The following section will discuss the two main directions researchers have focused on when evaluating green versus non-green investments, funds or stocks. Following on from that, the various models used to evaluate performance as evidenced in the literature, will be discussed. The remaining sections will discuss risk-adjusted returns, value/growth and cap-size investment styles in an attempt to justify this research's approach to performance evaluation and the categorisation of stocks within portfolios.

#### 2.3.1 Funds or Stocks

A mutual fund is a form of collective investments managed by a Fund Manager who trades the underlying securities on behalf of individual shareholders (Gandhi & Perumal, 2016). Equity funds, whereby managers maintain a diversified portfolio of equities

(stocks) for investors, are the largest category of mutual funds (Vyšniauskas & Rutkauskas, 2014) and typically come in various types reflecting the different types of stocks that the fund holds (see Figure 2.1). Categories of investment styles for stocks can be formed by a combination of the stock's market capitalisation (small, mid and large) and the stock's price-to-book ratio (value, growth or blend) (Liu & Wang, 2010a). When evaluating the performance of funds, it is common to classify the fund based on its investment practices and risk characteristics so that it can be measured against an appropriate benchmark that represents the style of the fund, typically an index (Dor et al., 2008).





Source: Vyšniauskas & Rutkauskas (2014)

The reviewed literature indicates that researchers focused mainly on the performance of funds versus the market as a whole or benchmarked against an index similar to the fund in question (Chang & Witte, 2010; Mallet & Michelson, 2010; Climent & Soriano,

2011; Sabbaghi, 2011; Chang et al., 2012; Muñoz, Vargas & Marco, 2012). Fund characteristics and returns are collected from databases such as Morningstar Principia (Chang & Witte, 2010; Mallet & Michelson, 2010; Chang et al., 2012; Muñoz et al., 2012) to analyse green or SRI funds categorised to an investment style such as large-cap blend, mid-cap value, small-cap growth etc., and compare with an average of Funds within the relevant categories (Chang et al., 2012) or against an index such as the S&P 500 Index (Mallet & Michelson, 2010). The usual periods of performance tested are one-year, three-year, five-year and ten-year with Chang et al. (2012) also testing 15-year returns. However, Mallet & Michelson (2010) note that as green funds are relatively new there is a small sample size of funds available with returns stretching back ten years.

In contrast, other researchers created sample portfolios of SRI or environmental stocks rather than use managed funds (Boulatoff & Boyer, 2009; Chia et al., 2009; Brzeszczyński & McIntosh, 2014; Lesser et al., 2014; Puopolo et al., 2015). However, the majority of research has focused on mutual funds rather than stocks (Lesser et al., 2014) thereby taking into account the fund manager's skill at picking stocks rather than evaluating the performance of green stocks in the market.

The outcome of this literature review is that a focus on stocks rather than funds would be a more suitable approach to examine the performance of green versus non-green investments, as this approach would take fund manager skill out of the equation. In this way, the performance of green stocks in the market can be evaluated against non-green stocks in the market without the impact of fund manager skill in the evaluation. The various approaches also illustrate the importance for this research to compare like with like when evaluating performance so that portfolios composed of green stocks of a particular investment style are evaluated against portfolios composed of non-green stocks of the same investment style.

#### 2.3.2 Measuring Returns

Various standard market sources can be used to collect performance returns. Lesser et al. (2014) used Thomson Reuters DataStream to collect monthly returns, whereas Sabbaghi (2011) collected price data from Yahoo!Finance to calculate daily returns for a five-year period for 15 green ETF funds identified through SustainableBusiness.com.

Risk-adjusted returns to measure the excess return (alpha) is the appropriate standard when comparing alternative investments (Climent & Soriano, 2011). Statistics from portfolio theory are used to calculate an investment's expected return based on its beta or risk measure, allowing researchers to compare risk adjusted returns with a benchmark (Chang et al., 2012) or using simple regression analysis to compare performance of stocks with company characteristics (Boulatoff & Boyer, 2009). Various models can be used to measure the expected return, the main ones in use are CAPM 1-factor (Chang & Witte, 2010; Chang et al., 2012), Fama-French 3-factor model, Carhart 4-factor model (Puopolo et al., 2015) and Barra Global Equity Model2 (Chia et al., 2009).

As can been seen from the literature therefore, a wide variety of methods and models are used to rank and evaluate performance of green or SRI investments. The following sections therefore will discuss the various methods and models used in an attempt to ascertain a suitable approach to evaluating the performance of green and non-green portfolios.

#### 2.3.3 Risk Adjusted Returns

Markowitz's (1952) Modern Portfolio Theory (MPT) assumes that markets are efficient and investors are risk averse, so when given two portfolios where expected returns are equal, they will favour the less risky portfolio (Lee, Cheng & Chong, 2016; Gasser, Rammerstorfer & Weinmayer, 2017). Based on MPT, the Capital Asset Pricing Model (CAPM) was developed by Sharpe, Litner and Mossin to explain the cross-section of stock returns, why one stock earns higher or lower returns to another (Fama & French, 2004). The assumption of CAPM is that there exists a relationship between market risk and expected returns (Lai & Stohs, 2015; Zaremba, 2016). The Sharpe ratio (Figure 2.2) is a practical method for classifying and measuring mutual fund performance, which measures the excess return of a fund or portfolio above the risk-free interest rate (Schröder, 2007; Statman & Glushkov, 2016).

Capelle-Blanchard & Monjon (2014) indicate that most academic research investigating SRI fund performance use the Sharpe ratio or a multifactor model such as the Fama-French three-factor model, to evaluate risk-adjusted returns. A high and positive Sharpe ratio indicates superior risk-adjusted returns whereas a low and negative ratio shows poor risk-adjusted returns (Gandhi & Perulam, 2016). The Fama-French three factor model is based on the claim that CAPM fails to take into account market cap size and price-to-book ratio, and is therefore considered a better tool for assessing portfolio performance (Panopoulou & Plastira, 2014). There are many more models that extend the Fama-French three-factor, but a three-factor model that includes cap-size and P/B ratio does as well as more elaborate methods (Chan, Hsiu-Lang & Lakonishok, 2002).

Figure 2.2: Sharpe Ratio

$$\mathrm{SR}_i = rac{\mu_i - r_f}{\sigma_i}$$

SR<sub>i</sub> = Sharpe ratio of investment portfolio,  $\mu_i$  = annualised return of investment portfolio,  $r_f$  = risk-free interest rate,  $\sigma_i$  = standard deviation of returns of investment portfolio **Source:** Schröder (2007)

This research will therefore use the Sharpe ratio when comparing the returns of green portfolios with non-green portfolios so as to evaluate the portfolios' risk-adjusted returns. As the Sharpe ratio is a measurement of the portfolio's excess return per unit of risk as defined by the portfolio's standard deviation, its use will enable a better comparison of portfolio performance on a risk-adjusted basis. Further, to take into account the elements of Fama-French three factor model, portfolios will be split by size and style. These elements will be discussed in the following two sections.

#### 2.3.4 Value or Growth (Price-to-book ratio)

Value and growth are two opposing investment styles where investors consider value stocks to be under-valued by the market, whereas growth stocks, not usually under-valued, are considered to have strong growth potential (Hodnett & Hsieh, 2012). Price-to-book (P/B) ratio is the market value of a firm's equity divided by the book value of its equity (Nezlobin, Rajan & Reichelstein, 2016). The P/B ratio can be used to classify stocks as either value or growth, where stocks with a low P/B ratio been defined as value stocks, conversely stocks with high P/B ratios been defined as growth stocks (Bauer, Denva & Otten, 2006; Kempf & Osthoff, 2007; Kim & Mulvey, 2009).

The Fama-French three-factor model (see Figure 2.3) attempts to account for the return variance between value and growth stocks and large and small stocks (Bauer et al., 2006). Its premise being that CAPM fails to take into account cross-sectional variation of stock returns, and therefore the three-factor model expands CAPM by including the return difference between a portfolio of small-cap stocks and a portfolio of large-cap stocks (SMB) and the return difference between a portfolio of stock premium (MRP) (Pratt, 2011; Dolinar, 2013; Panopoulou & Plastira, 2014; Vo, 2015).

#### Figure 2.3: Fama-French three factor model

$$R_{p,t} - R_{f,t} = \alpha_p + \beta (R_{m,t} - R_{f,t}) + s \times SMB_t + h \times HML_t$$

1

where:

*R*<sub>p,t</sub> - *R*<sub>f,t</sub> represents an excess return for a portfolio;

•  $(R_{m,t} - R_{f,t})$  is the MRP; and

 β, s, h are the estimated coefficients for the MRP; the mimicking size portfolio and the mimicking book-to-market portfolios.

**Source:** Vo (2015)

The available literature indicates that researchers employing the Fama-French threefactor model have found evidence that there are performance differences between growth and value portfolios. Borys & Zemčik (2011) constructed portfolios based on P/B and cap-size to demonstrate that markets in Eastern European countries are similar to US and other developed world markets in relation to size and value explaining expected returns. Research by Capual, Rowley & Sharpe (1993) demonstrated that portfolios composed of low P/B ratios (value stocks) outperformed portfolios composed of high P/B ratios (growth stocks) over a ten-year period. Later, studies by Bauman, Conover & Miller (1998), covering 28,000 annual stock returns found that value stocks outperformed growth stocks in both total and risk-adjusted returns over a ten-year period. More recently, Liu & Wang's (2010b) research, shows that in the short term, value stocks have greater risk and returns in comparison to growth stocks, but lower risk and greater returns in the long term. In contrast, Chan et al. (2002) found that growth fund managers outperform their value counterparts by 1.2% per year on average.

This research will therefore split stocks into separate portfolios by using the P/B ratio to identify value and growth stocks. This will enable green value portfolios to be evaluated against non-green value portfolios, and green growth portfolios to be evaluated against non-green growth portfolios. Therefore, the HML element of the Fama-French three-factor model will be catered for in the portfolio split and will not need to be included in the return calculations.

#### 2.3.5 Market Cap Size

Market capitalization is calculated by multiplying the price of a single share of a company's stock by the number of outstanding shares. Small-cap stocks typically carry greater risk than mid or large-cap, and although they are often able to increase earnings more rapidly than larger companies they also tend to fall harder in a bear market (Eisinberg, 2000). A large-cap firm is one with a market cap of greater than \$10 billion, mid-cap between \$2 and \$10 billion and small cap below \$2 billion (Zacks, 2014).

Investing at specific times in either value or growth stocks or large cap or small cap is a type of "style investing" that can be an important strategy for an investor to maximise returns (Moerloose & Giot, 2011). Kim & Mulvey (2009) note that many researchers have found a size effect where market capitalization can account for cross-sectional expected returns, with small stocks performing better and large stocks performing worse than the CAPM predicted returns. Liu & Wang's (2010a) research show that large-cap growth style is the least risky over the short term but small-cap value style is the least risky for longer term investments, with small-growth being the riskiest style. The most advanced type of research on comparisons of SRI and non-SRI funds, employ a like-for-like matching approach, where funds with similar characteristics such as size and style are compared (Schröder, 2007).

This research will therefore further split stocks into separate portfolios by market capsize. This will enable green portfolios to be evaluated against non-green portfolios based on their corresponding cap size. Therefore, the SMB element of the Fama-French three-factor model will be catered for in the portfolio split and will not need to be included in the return calculations.

#### 2.4 How have SRI and green stocks measured up?

This section will discuss the findings of the reviewed literature with regards to how investment performance of SRI and green investments have compared with more traditional investments.

#### 2.4.1 Lack of Consensus

Most research in the evaluation of performance between SRI and non-SRI investing has shown no significant difference between the two. Studies (Mallet & Michelson, 2010; Climent & Soriano, 2011) have shown that divergence of risk adjusted returns between

green, SRI and conventional funds is not significant. In a recent paper reviewing research on SRI, Junkus & Berry (2014), found most studies reported no significant difference between SRI performance and traditional investing. This is further backed up by Przychodzen et al. (2016) who argue that the existing literature offers no consensus on a correlation of investment performance with a policy of incorporating ESG into an investment strategy.

Other studies have shown differences between green and non-green stocks versus a market index, however results are not necessarily in agreement. For example, research concentrating on renewable energy (Chia et al, 2009) display statistically significant superior performance of international green stocks versus the MSCI All Country World Index, in contrast Boulatoff & Boyer (2009) reveal that the Nasdaq outperforms international environmental stocks.

Differences in performance have been explained in some instances by the impact of market cycles. Muñoz et al. (2014) found that green and SRI funds relative performance has been statistically insignificant to the market in times of crisis, but underperform in normal periods. Lesser et al. (2014), extending Climent & Soriano (2011) studies on US Environmental Funds to international markets, found that green outperformed SRI between 2003 and 2007 and underperformed between 2008 and 2012, owing mainly to particular portfolio weighting, leading the researchers to conclude that green investing can be considered a sector bet on renewable energies.

Other research shows differences in portfolio performance depending on investment styles. Chang & Witte (2010) show that US SRI funds have inferior returns to the average return of funds within the same categories with the exception of mid-cap value funds and small-cap blend funds. Chang et al. (2012) show that US green mutual funds in the main underperform, displaying lower risk-adjusted returns than category averages over 5-Year and 10-Year periods, with results over 3-year and 15-year being statistically insignificant. However, areas where green funds outperformed the category averages

were in large blend and mid-cap blend over five and ten years and large growth over a ten-year period. These longer time frames and investment styles therefore, are areas that this study would look at to determine if green stocks can show superior performance.

In an attempt to overcome the lack of consensus in previous literature, Reveilli & Viviani (2015) undertook a meta-analysis study of 85 previous studies and 190 experiments and found that there was no real cost or benefit to investing in SRI and that any performance differences by previous researchers were mainly due to methodological choice or the management skills of specific fund managers to outperform the market. Further, Muñoz et al. (2014) posit that an explanation for the differences between their findings and those of Climent & Soriano (2011) may be explained by the different periods under consideration and the different sample of funds.

Ideally therefore, research of this nature would include a period covering as many crisis/non-crisis periods as possible. By examining peak and troughs of the S&P 500 Index, Nofsinger & Vargo (2014) identify two stock market crisis periods: the 2001 dotcom collapse and the 2008 financial crisis. Petajisto (2013), show how different categories of mutual funds performed over the two year-period 2008-2009, indicating that subsequent to a financial crash, different investment styles can affect investment performance over the market. Accordingly, for this research, a deeper analysis would be possible if environmental ratings of stocks could be analysed over as great a time frame as possible, ideally prior to the 2001 dot-com bubble.

#### 2.5 Conclusion & Gap in Literature

As can be seen from the existing body of literature, the many measurements and time periods under investigation fail to clearly identify trends. Although some research concentrates on the performance of green stocks, the main focus of the reviewed literature has been the performance of green funds in comparison to either SRI or to the market as a whole. The literature has also identified a challenge in categorizing funds as green or non-green. Additionally, given that green funds are relatively new, sampling enough green funds over a long enough period to detect trends is also a challenge. A further challenge, identified in the literature, with using funds to analyse trends in green investing is the factor of fund manager skill.

This study therefore, should evaluate stocks over as great a timeframe as possible facilitating the evaluation of investment performance over several crisis/non-crisis periods. Also, to remove the element of skill of fund managers in picking stocks, portfolios of stocks selected from a market index should be used to evaluate the behaviour of green stocks rather than evaluating funds. This will have the benefit of being able to evaluate the performance of green stocks in the market, as opposed to evaluating the performance of fund managers. Additionally, using environmental pillar scores from MSCI ESG research to identify what is green and non-green would be a more objective approach to the research than attempting to pick a sample of green stocks or funds. Portfolios should be further split to cater for cap-size and value/growth characteristics to cater for the SMB and HML elements of the Fama-French three-factor model. This will enable a cross-sectional comparison of a portfolio composed of green stocks versus one composed of non-green stocks.

The literature review has helped steer the focus of this study and enabled a clear research aim to be established. The research aim and objectives will be discussed in the next section.

17

## **3** Research Question

#### 3.1 Research Aims & Objectives

The purpose of the dissertation is to construct and analyse portfolios comprised of green and non-green stocks, based on the MSCI ESG Environmental pillar scores and categorised within appropriate investment styles, with the objective of comparing their performances over 3-Year, 5-Year and 10-Year periods, and over the 2-Year crisis-period 2008-2009.

The importance of the research is to discover trends in green investment performance which may be informative and helpful for investors when determining their stock selection strategies.

#### 3.2 Hypotheses

Having reviewed the published literature in this field, the following hypotheses have been developed:

Hypothesis 1

H0: There is no difference between the investment performance of a growth/large-cap styled portfolio composed of green stocks and a growth/large-cap styled portfolio composed of non-green stocks over the last 3 years?

#### Hypothesis 2

H0: There is no difference between the investment performance of a growth/large-cap styled portfolio composed of green stocks and a growth/large-cap styled portfolio composed of non-green stocks over the last 5 years?

#### Hypothesis 3

H0: There is no difference between the investment performance of a growth/large-cap styled portfolio composed of green stocks and a growth/large-cap styled portfolio composed of non-green stocks over the last 10 years?

#### Hypothesis 4

H0: There is no difference between the investment performance of a growth/large-cap styled portfolio composed of green stocks and a growth/large-cap styled portfolio composed of non-green stocks over the two-year crisis-period of 2008 and 2009?

#### Hypothesis 5

H0: There is no difference between the investment performance of a value/large-cap styled portfolio composed of green stocks and a value/large-cap styled portfolio composed of non-green stocks over the last 3 years?

#### Hypothesis 6

H0: There is no difference between the investment performance of a value/large-cap styled portfolio composed of green stocks and a value/large-cap styled portfolio composed of non-green stocks over the last 5 years?

#### Hypothesis 7

H0: There is no difference between the investment performance of a value/large-cap styled portfolio composed of green stocks and a value/large-cap styled portfolio composed of non-green stocks over the last 10 years?

#### Hypothesis 8

H0: There is no difference between the investment performance of a value/large-cap styled portfolio composed of green stocks and a value/large-cap styled portfolio composed of non-green stocks over the two-year crisis-period of 2008 and 2009?

#### Hypothesis 9

H0: There is no difference between the investment performance of a growth/mid-cap styled portfolio composed of green stocks and a growth/mid-cap styled portfolio composed of non-green stocks over the last 3 years?

#### Hypothesis 10

H0: There is no difference between the investment performance of a growth/mid-cap styled portfolio composed of green stocks and a growth/mid-cap styled portfolio composed of non-green stocks over the last 5 years?

#### Hypothesis 11

H0: There is no difference between the investment performance of a growth/mid-cap styled portfolio composed of green stocks and a growth/mid-cap styled portfolio composed of non-green stocks over the last 10 years?

#### Hypothesis 12

H0: There is no difference between the investment performance of a growth/mid-cap styled portfolio composed of green stocks and a growth/mid-cap styled portfolio composed of non-green stocks over the two-year crisis-period of 2008 and 2009?

#### Hypothesis 13

H0: There is no difference between the investment performance of a value/mid-cap styled portfolio composed of green stocks and a value/mid-cap styled portfolio composed of non-green stocks over the last 3 years?

#### Hypothesis 14

H0: There is no difference between the investment performance of a value/mid-cap styled portfolio composed of green stocks and a value/mid-cap styled portfolio composed of non-green stocks over the last 5 years?

#### Hypothesis 15

H0: There is no difference between the investment performance of a value/mid-cap styled portfolio composed of green stocks and a value/mid-cap styled portfolio composed of non-green stocks over the last 10 years?

#### Hypothesis 16

H0: There is no difference between the investment performance of a value/mid-cap styled portfolio composed of green stocks and a value/mid-cap styled portfolio composed of non-green stocks over the two-year crisis-period of 2008 and 2009?

The following chapter will discuss the methodology used to test the hypotheses.
# 4 Methodology

The research seeks to understand if portfolios of green stocks, in different investment style categories, can outperform portfolios of non-green stocks when evaluated within the same investment style categories. This research used a method to rebalance portfolios similar to one used in a study on the effect of socially responsible investing on portfolio performance (Kempf & Osthoff, 2007). Kempf & Osthoff (2007) constructed two portfolios, by taking an SRI rating at the end of each year to determine if a stock was to be classified in the SRI or non-SRI portfolio for the following year, rebalancing every year, and then generating a time series of monthly returns for a 12-year period for the two portfolios.

For this research, therefore, portfolios of stocks from the MSCI All World Index were constructed using the MSCI ESG Research environmental pillar ratings, stock style (growth or value) and cap-size (large and small). Portfolios were rebalanced for each of the ten-year sub-periods. A time-series of monthly returns was generated for each of the ten-year sub-periods for the portfolios constructed. Periodic (3-year, 5-year, 10-year and 2008-2009 crisis period) returns of each green portfolio were compared with their non-green equivalent.

This chapter will outline the methodological choices encountered, any assumptions made, data collection, data analysis and statistical analysis undertaken. SQL procedures and tables were used in the data analysis and are available on request from the author.

# 4.1 Methodological Choice

Quantitative research implies quantification in data collection and analysis, requiring a process of deduction adopting practices of the natural scientific model, positivism and objectivism. Qualitative research on the other hand is often associated with interpretive

philosophy, where data collection is often non-standard and employing non-probability sampling techniques (Bryman & Bell, 2011).

Research Philosophy relates to the development and nature of knowledge with epistemology, ontology and axiology being three major philosophical research assumptions (Bryman & Bell, 2011; Quinlan, 2015; Saunders et al., 2015). This paper's research is concerned with data collection and objective analysis, focusing on facts rather than impressions and therefore will take a positivist epistemological position rather than realism or interpretivism one. The research seeks to identify and compare the investment performance of green and non-green portfolios and the results are not dependent on the researcher's view of reality. This study's ontological position therefore, is one of objectivism rather than subjectivism. This research evaluates investment performance from existing secondary data and is carried out independently of the values of the researcher who should be detached, neutral and independent. From an axiology view point therefore, the research is carried out in a value-free manner.

This research collects and objectively analyses numeric data to evaluate green investing, thereby taking a positivist epistemological position, implying quantitative research (Saunders et al. 2015). The evaluation of performance returns of stocks based on their investment style, size and historic MSCI environmental ratings is objective in nature and concerned with numbers rather than words. The methodological choice most suitable therefore, is a quantitative one and is in keeping with studies of this nature (Boulatoff & Boyer, 2009; Sabbaghi, 2011; Chang & Witte, 2012; Lesser et al., 2014;).

23

# 4.2 Data Collection

# 4.2.1 Stocks or Funds

This research compared the performance of green stocks versus non-green stocks in the market as opposed to the performance of green funds versus non-green funds. The majority of research undertaken in this field thus far has focused on mutual funds rather than stocks, thereby taking into account fund manager skills at picking stocks, rather than evaluating the performance of green stocks in the market (Lesser et al., 2014). Other research however, has created sample portfolios of SRI or environmental stocks rather than use managed funds (Schröder, 2007; Boulatoff & Boyer, 2009; Chia et al., 2009; Brzeszczyński & McIntosh, 2014; Lesser et al., 2014; Puopolo et al., 2015). Schröder (2007) argues that concentrating on the constituents of indices rather than funds removes obscurities such as transaction costs, management skills and timing activities of the fund manager which are not relevant to the question at hand.

This research therefore adopted the method of building sample portfolios of green and non-green stocks rather than use managed funds. With this approach, the performance of stocks over the relevant periods were evaluated for the full period, as stocks were not removed from a portfolio based on a fund manager's stock selection skill, nor were there any transaction costs or management fees associated with the portfolio performance.

# 4.2.2 MSCI Index

The use of the constituents of indices to identify and classify market stocks is the norm in research papers of this nature. Brzeszczyński & McIntosh (2014) use the FTSE100 index and FTSE4GOOD index as benchmarks to compare with the performance of portfolios composed of British SRI stocks. In other research, Zaremba (2016), examining the relationship between risk and return of stocks based on international stock markets, argues that the selection of MSCI indices is justified as it aligns the research with investment practice. Zaremba (2016) reasons that MSCI indices are maintained with the purpose of being fully investable from an international perspective and contain approximately 85% of all stock market capitalizations globally. Additionally, MSCI is considered one of the top three financial service agencies that provide ESG scores (Tarmuji et al., 2016). Use of the MSCI ACWI in this study is valid as it is in keeping with previous research of this nature. The MSCI ACWI is an index of the global stock market and currently contains over 2,400 large and mid-cap stocks with a geographic reach across both developed and emerging investment markets.

The constituents of MSCI ACWI along with monthly P/B ratio, market capitalisation and environmental ratings were downloaded to three separate Excel files from FactSet Research Systems, a licensed provider of MSCI historic data (MSCI, 2017a). The Excel files were uploaded to three SQL tables for further manipulation, which will be discussed in the following section on Data Analysis.

# 4.3 Data Analysis

# 4.3.1 Portfolio Rebalancing

The data collection facilitated the construction of green and non-green portfolios of four investment styles: large-cap growth, large-cap value, mid-cap growth and mid-cap value. There were therefore eight portfolios (Figure 4.1) constructed based on investment style, cap-size and environmental ratings.

Portfolio	Portfolio Description
GGL	Green Growth Large-Cap
GGM	Green Growth Mid-Cap
GVL	Green Value Large-Cap
GVM	Green Value Mid-Cap
NGL	Non-Green Growth Large-Cap
NGM	Non-Green Growth Mid-Cap
NVL	Non-Green Value Large-Cap
NVM	Non-Green Value Mid-Cap

Table 4.1: Green and Non-Green Portfolios

Portfolio rebalancing refers to the need to maintain and re-adjust the constituents of a portfolio's stock allocation to keep it in line with its original strategic allocation (Kohler & Wittig, 2014). The constituents of the MSCI ACWI index and the classifications of stocks are not static. A stock's environmental rating changes and may change enough for it to be reclassified from green to non-green or vice-versa. A stock's investment style may change from value to growth or vice versa. The market capitalisation of a stock changes as its share price or share issuance changes, and may change enough for it to be reclassified from large-cap to mid-cap or vice versa. Therefore, each portfolio was required to be reconstructed periodically to allow for any changes to the MSCI ACWI index constituents, classification and environmental rating.

Each of the eight portfolios were rebalanced yearly rather than monthly for the ten annual sub-periods from 2007 to 2016. Portfolios were rebalanced based on the previous year-end value for each of the three categories of cap-size, P/B ratio and environmental ratings stored in the three SQL tables. This therefore, meant that there were 80 portfolio/year combinations for which investment performance returns were required, as opposed to 960 portfolio/month combinations that would have been required if portfolios were rebalanced on a monthly basis. This not only ensured that the data analysis was more manageable, but was also in keeping with the methods applied in research of this type (Brzeszczyński & McIntosh, 2014).

The starting position for environmental ratings for any given year was based on the last environmental rating for each stock from the previous year. The starting position for cap-size for any given year was based on the last cap-size for each stock from the previous year. The starting position for P/B for any given year was based on the last P/B for each stock from the previous year.

To rebalance the portfolios, a cut-off point was calculated for environmental ratings and value/growth style for each year. The cap-size cut off point to determine mid-cap or large-cap remains the same at \$10 billion, for each year. There are no small-cap stocks in the MSCI ACWI. Stocks with a cap-size below \$10 billion are mid-cap, and above \$10 billion are large-cap (Zacks, 2014). This research therefore classified any stock with cap-size of \$10 billion or larger as large-cap and any stock with a cap-size of less than \$10 billion as mid-cap. This classification was used in determining which portfolio a stock will be assigned to for each year's rebalancing.

To calculate a value/growth style cut-off, a dividing line based on P/B ratio can be calculated to determine whether a stock is a value stock or a growth stock (Capual, Rowley & Sharpe, 1993). Anything above the dividing line is considered a growth stock, anything below the line is considered a value stock. Fisher (1992) defines stocks above the median P/B at any given point in time as a growth stock and anything below the median as a value stock. For this research, the median P/B ratio for each of the 10 yearly sub-periods was calculated at the end of each year and used as the cut-off point for classifying stocks as either growth or value for the following year. This classification was used in determining which portfolio a stock was assigned to for each year's rebalancing.

To calculate an environmental rating cut off point, a similar approach to the Filbeck et al. (2014) method is used, the top 30% and bottom 30% of stock environmental scores

from the MSCI ESG ratings determines if a stock is classified as green or non-green. For this research, top 30% and bottom 30% cut-off points for each of the 10 yearly subperiods were calculated at the end of each year and used as the cut-off point for classifying stocks for the following year, with any stock in the top 30% classified as green and any stock in the bottom 30% classified as non-green. This classification was used to determine which portfolio a stock was assigned to for each year's rebalancing. Figure 4.2 below displays a table listing of the number of stocks in each rebalanced portfolio for each year.

Portfolio	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016
GGL	150	176	147	147	147	133	187	263	268	280
GGM	82	80	123	97	79	97	158	167	167	193
GVL	161	143	65	108	144	117	133	142	158	137
GVM	158	150	209	155	136	168	207	175	173	174
NGL	133	139	65	113	129	121	149	136	147	103
NGM	175	161	215	158	124	130	193	206	181	209
NVL	96	86	35	62	83	65	101	108	111	93
NVM	149	162	229	175	168	193	255	325	346	387

Table 4.2: Number of stocks per portfolio per year

#### 4.3.2 Generate Returns

Each Portfolio was uploaded to Thomson Reuters Eikon to generate monthly returns for 2007. The portfolios were then reconstructed and reloaded to get monthly returns for the following year and so on until returns for all portfolios up to 2016 were retrieved, giving a total of 120 monthly returns for each of the eight portfolios. Thomson Reuters Eikon is an industry standard tool used to monitor and analyse financial information. The reliability of Thomson Reuters Eikon has not been questioned in the academic or

corporate community (Garcia, Mendes-Da-Silva & Orsato, 2017), and so was an appropriate choice of tool for research of this nature.

The monthly returns were statistically analysed to see how each green portfolio based on style, size and environmental ratings compared with its non-green counterpart. Independent-sample t-tests are used to compare the mean score on a continuous variable for two groups (Pallant, 2016) and are based on the assumptions of level of measurement, random sampling, independence of observations, normal distribution and homogeneity of variance. Independent-sample t-tests therefore, were appropriate for this research as each green portfolio was evaluated against one other portfolio i.e. its non-green equivalent. Levene's test for equality of variance was used to determine which result-set to use from the Independent-sample t-tests.

Shapiro Wilk test was used to test for normal distribution between each group. Mann-Whitney U Test is a non-parametric alternative to the t-test where the distributions are not normal (Pallant, 2016) and so was used in this research for any test where one of the groups being tested had a p<0.05 in the Shapiro Wilk test.

Similar to the Brzeszczyński & McIntosh (2014) approach, the time-series of monthly returns were chain linked to calculate an annualised geometric time weighted returns for the 3-year, 5-year and 10-year periods for the purpose of comparison between different periods. As the geometric method, also called time-weighted return (see Figure 4.1), does not ignore compounding, it is a preferred measurement to the arithmetic method when evaluating past performance (Lee, 2012). Time weighted-returns are an appropriate choice therefore in this research for investment performance evaluation.

#### Figure 4.1: Time Weighted Returns

$$TWR = \left(\prod_{t=1}^T \left(1+r_t\right)\right)^{1/T} \! - \! 1$$

TWR is the time-weighted return,  $r_t$  is the return in period t and T is the number of time periods.

**Source:** Lee (2012)

Sharpe ratios were calculated for each portfolio over 3-year, 5-year, 10-year and 2008-2009 sub-periods. Explica investment analysis software was used to calculate Sharpe ratios, using the 3-month Inter-bank Libor EUR rate as the risk-free rate. Explica software is used by financial companies (Royal Bank of Scotland, AIB, Bank of Ireland, AXA and Bloxham) to generate a range of statistical analyses (Enterprise Ireland, 2006), and so was a suitable choice of tool for research of this nature. Using 3-month rates as the risk-free rate to calculate Sharpe ratios is common practice in research of this nature (Brzeszczyński & McIntosh, 2014; Gang & Qian, 2016) and so was appropriate for this research.

# 4.3.3 Limitations

The MSCI ACWI is an index of the global stock market and currently contains over 2,400 large and mid-cap stocks with a geographic reach across both developed and emerging investment markets. Small-cap stocks are available in the MSCI All Country World Small Cap Index, however, the environmental ratings from MSCI ESG Research do not cover the small cap index. Additionally, there were no environmental ratings available for stocks prior to 2007 via MSCI ESG research. This highlights two limitations to the study, first, the exclusion of small-cap stocks from the research due to the lack of environmental ratings for small-cap stocks, and second, the timeframe is limited to a ten-year period from 2007 to 2016, thereby excluding the 2001 dot-com collapse. This meant that only one crisis period, the 2008 financial crisis, was covered by the research and so the results can not imply any particular trend in the investment performance

returns subsequent to a market crash. Additionally, no conclusions regarding the performance of green small stocks versus non-green small stocks can be drawn given the absence of small stocks from the research.

# 4.4 Methodology Summary

The MSCI ESG environmental ratings, market cap-size and P/B ratio of stocks in the MSCI ACWI were used to compile four green portfolios and four non-green portfolios. The eight portfolios were reconstructed for each year from 2007 to 2016 based on a stock's environmental rating, cap-size and P/B ratio as of the end of the previous year. Thomson Reuters Eikon was used to generate monthly performance returns for each portfolio for each year. The monthly investment returns of each green portfolio for each year. The monthly investment returns of each green portfolio for each user compared to its non-green equivalent over 3-Year, 5-Year, 10-Year and the 2008-2009 periods. A geometric time weighted return was also computed and annualised for each portfolio and sub-period for further analysis. Additionally, a Sharpe ratio was computed to compare risk-adjusted returns.

The following chapter will present the findings of the research.

# **5** Research Findings

# 5.1 Introduction

This section outlines the analysis and findings of the performance returns of green portfolios compared to non-green portfolios across the four styles of large/growth, large value, mid/growth and mid/value. The four sub-periods, 3-year, 5-year, 10-year and 2008-2009 geometric returns for each style will be detailed graphically. The Sharpe ratio of each portfolio in each of the four sub-periods will also be detailed.

The purpose of the dissertation was to construct and analyse portfolios comprised of green and non-green stocks based on the MSCI ESG Environmental pillar scores and categorised within appropriate investment styles with the objective of comparing their performance with each other over 3-Year, 5-Year and 10-Year periods, and over the 2-Year crisis-period 2008-2009.

# 5.2 Hypotheses Testing

The monthly returns were analysed using the statistical software package SPSS. Tests were carried out corresponding to each of the 16 hypotheses. Each test was carried out to determine differences between the means of two independent groups, a green and non-green group. A Shapiro Wilk's test of normality was carried out between each group to determine which test to use. An independent Samples t-Test was used where Shapiro Wilk's test found no significant deviations from normality for both groups. A Mann-Whitney U-Test was carried out as a non-parametric alternative to the Independent Samples t-Test. The p-value is the Sig. (2-tailed) of each test and is used to determine statistically significant differences between the means of two groups. A p-value of less than or equal to 0.05 indicates a statistically significant difference between the means of the two groups. A p-value of greater than 0.05 will indicate no statistically significant difference between the means of the two groups.

#### 5.2.1 Growth / Large-Cap 3-Year Returns

Hypothesis 1, H0: There is no difference between the investment performance of a growth/large-cap styled portfolio composed of green stocks and a growth/large-cap styled portfolio composed of non-green stocks over the last 3 years?

The null hypothesis for this test is that there is no statistically significant difference between the tested groups.

#### 5.2.1.1 Test of Normality

The sample data for the 3-year returns for the groups Green Growth Large-Cap (GGL) and Non-Green Growth Large-Cap (NGL) were first analysed to test that each group was normally distributed. The null hypothesis for this test is that the data sample is normally distributed and the alternative hypothesis is that the data is not normally distributed.

**Table 5.1**: Test of normality for Green and Non-Green Growth/Large Cap Portfolios 3-Year

	-										
		Kolm	ogorov-Smir	nov <sup>a</sup>	Shapiro-Wilk						
	PortfolioType	Statistic	df	Sig.	Statistic	df	Sig.				
PortfolioReturns	GGL	.130	36	.131	.977	36	.628				
	NGL	.141	36	.068	.967	36	.348				

Tests of Normality

a. Lilliefors Significance Correction

Table 5.1 shows that the results of the Shapiro Wilk test of normality for GGL and NGL for the 3-Year period show that there were no significant deviations from normality  $(W_{GGL} = .977, df = 36, p = .628), (W_{NGL} = .967, df = 36, p = .348).$ 

As both p-values are higher than 0.05 the null hypothesis is not rejected and therefore given the lack of identified deviations in normality, an Independent Samples t-Test can be relied upon to test if there are significant differences between the monthly returns of GGL and NGL in the 3-Year sub-period.

# 5.2.1.2 Test of Significance

The result of Levene's test for equality of Variance (Table 5.3) shows that there is insufficient evidence to reject the assumption of equal variances and so equal variances are assumed (**F** = **.156**, **p** = **.694**). The results of the Independent Samples t-Test (Table 5.3) show there is insufficient evidence to suggest that the monthly returns are different between the GGL portfolio (**M**=**1.07**, **SD**=**3.33**, **n**=**36**) and the NGL portfolio (**M**=**.85**, **SD**=**3.17**, **n**=**36**) (Table 5.4) over a 3-Year period (**t** (**70**) = **.290**, **p** = **.773**), therefore, the research fails to reject the null hypothesis.

# Table 5.2: Groups Statistics – GGL versus NGL 3-Year period

Group Statistics									
	PortfolioType	Ν	Mean	Std. Deviation	Std. Error Mean				
PortfolioReturns	GGL	36	1.0727553	3.33315604	.55552601				
	NGL	36	.8505314	3.16796790	.52799465				

# Table 5.3: Independent Samples t-Test – GGL versus NGL 3-Year period

		Levene's Test Varia	Levene's Test for Equality of Variances ttest for Equality of Means							
							Mean	Std. Error	95% Confidence Interval of the Difference	
		F	Sig.	t	df	Sig. (2-tailed)	Difference	Difference	Lower	Upper
PortfolioReturns	Equal variances assumed	.156	.694	.290	70	.773	.22222389	.76641209	-1.30633683	1.75078461
	Equal variances not assumed			.290	69.820	.773	.22222389	.76641209	-1.30640615	1.75085392

#### Independent Samples Test

# 5.2.2 Growth / Large-Cap 5-Year Returns

Hypothesis 2, H0: There is no difference between the investment performance of a growth/large-cap styled portfolio composed of green stocks and a growth/large-cap styled portfolio composed of non-green stocks over the last 5 years?

The null hypothesis for this test is that there is no statistically significant difference between the tested groups.

#### 5.2.2.1 Test of Normality

The sample data for the 5-year returns for the groups Green Growth Large-Cap (GGL) and Non-Green Growth Large-Cap (NGL) were analysed to test that each group was normally distributed. The null hypothesis is that the data sample is normally distributed and the alternative hypothesis is that the data is not normally distributed.

 Table 5.4: Test of normality for Green and Non-Green Growth/Large-Cap Portfolios 5 

 Year

	rests of normality										
		Kolm	ogorov-Smiı	rnov <sup>a</sup>	Shapiro-Wilk						
	PortfolioType	Statistic	df	Sig.	Statistic	df	Sig.				
PortfolioReturns	GGL	.081	60	.200	.976	60	.294				
	NGL	.117	60	.041	.968	60	.114				

Tests of Normality

\*. This is a lower bound of the true significance.

a. Lilliefors Significance Correction

Table 5.5 shows that the results of the Shapiro Wilk test of normality for GGL and NGL for the 5-Year period show that there were no significant deviations from normality ( $W_{GGL} = .976$ , df = 60, p = .294), ( $W_{NGL} = .968$ , df = 60, p = .114).

As both p-values are higher than 0.05 the null hypothesis is not rejected and therefore given the lack of identified deviations in normality, an Independent Samples t-Test can be relied upon to test if there are significant differences between the monthly returns of GGL and NGL in the 5-Year sub-period.

# 5.2.2.2 Test of Significance

The result of Levene's test for equality of Variance (Table 5.6) shows that there is insufficient evidence to reject the assumption of equal variances and so equal variances are assumed ( $\mathbf{F} = .109$ ,  $\mathbf{p} = .742$ ). The results of the Independent Samples t-Test (Table 5.6) show there is insufficient evidence to suggest that the monthly returns are different between the GGL portfolio (M=1.19, SD=2.89, n=60) and the NGL portfolio (M=.94,

**SD=2.83**, **n=60**) (Table 5.5) **over** a 5-Year period **(t (118) = .5, p = .618).** Therefore, the research fails to reject the null hypothesis.

Table 5.5: Groups Statistics – GGL versus NGL 5-Year period

or oup statistics									
	PortfolioType	Ν	Mean	Std. Deviation	Std. Error Mean				
PortfolioReturns	GGL	60	1.1989175	2.89611193	.37388644				
	NGL	60	.9375393	2.83045341	.36540996				

Table 5.6: Independent Samples t-test – GGL versus NGL 5-Year period

# 5.2.3 Growth / Large-Cap 10-Year Returns

Hypothesis 3, H0: There is no difference between the investment performance of a growth/large-cap styled portfolio composed of green stocks and a growth/large-cap styled portfolio composed of non-green stocks over the last 10 years?

	Independent Samples Test												
Levene's Test for Equality of Variances					t-test for Equality of Means								
							Mean	Std. Error	95% Confidence Differ	e Interval of the ence			
		F	Sig.	t	df	Sig. (2-tailed)	Difference	Difference	Lower	Upper			
PortfolioReturns	Equal variances assumed	.109	.742	.500	118	.618	.26137817	.52279586	77389997	1.29665631			
	Equal variances not assumed			.500	117.938	.618	.26137817	.52279586	77390561	1.29666194			

The null hypothesis for this test is that there is no statistically significant difference between the tested groups.

## 5.2.3.1 Test of Normality

The sample data for the 10-year returns for the groups Green Growth Large-Cap (GGL) and Non-Green Growth Large-Cap (NGL) were analysed to test that each group was normally distributed. The null hypothesis is that the data sample is normally distributed and the alternative hypothesis is that the data is not normally distributed.

**Table 5.7**: Test of normality for Green and Non-Green Growth/Large Cap Portfolios 10-Year

	reas of Normality											
		Kolm	ogorov-Smir	nov <sup>a</sup>	Shapiro-Wilk							
	PortfolioType	Statistic	df	Sig.	Statistic	df	Sig.					
PortfolioReturns	GGL	.093	120	.013	.962	120	.002					
	NGL	.129	120	.000	.957	120	.001					

Toete of Normality

a. Lilliefors Significance Correction

Table 5.7 shows that the results of the Shapiro Wilk test of normality for GGL and NGL for the 10-Year period show that there were significant deviations from normality ( $W_{GGL}$  = .962, df = 120, p = .002), ( $W_{NGL}$  = .957, df = 120, p = .001).

As both p-values are less than 0.05 the null hypothesis is rejected and therefore as a deviation from normality is indicated, a Mann Whitney U-test is used to test if there are significant differences between the monthly returns of GGL and NGL in the 10-Year subperiod.

# 5.2.3.2 Test of Significance

The results of the Mann Whitney U-test (Tables 5.8 and 5.9) show there is insufficient evidence to suggest that the monthly returns are different between the GGL portfolio with a mean rank score of **121.39** and the NGL portfolio with a mean rank score of **119.61** over a 10-Year period (**U=7093**, **p=.842**). The research therefore, fails to reject the null hypothesis.

Table 5.8: Mean Rank – GGL versus NGL 10-Year period

Ranks

	PortfolioType	Ν	Mean Rank	Sum of Ranks
PortfolioReturns	GGL	120	121.39	14567.00
	NGL	120	119.61	14353.00
	Total	240		

## Table 5.9: Mann Whitney U test – GGL versus NGL 10-Year period

# Test Statistics<sup>a</sup>

	PortfolioRetur ns
Mann-Whitney U	7093.000
Wilcoxon W	14353.000
Z	199
Asymp. Sig. (2-tailed)	.842

a. Grouping Variable: PortfolioType

# 5.2.4 Growth / Large-Cap 2008-2009

Hypothesis 4, H0: There is no difference between the investment performance of a growth/large-cap styled portfolio composed of green stocks and a growth/large-cap styled portfolio composed of non-green stocks over the two-year crisis-period of 2008 and 2009?

The null hypothesis for this test is that there is no statistically significant difference between the tested groups.

# 5.2.4.1 Test for Normality

The sample data for the 2008-2009 returns for the groups Green Growth Large-Cap (GGL) and Non-Green Growth Large-Cap (NGL) were analysed to test that each group was normally distributed. The null hypothesis is that the data sample is normally distributed and the alternative hypothesis is that the data is not normally distributed. **Table 5.10**: Test of normality for Green and Non-Green Growth/Large Cap Portfolios 2008-2009

		Kolm	ogorov-Smii	rnov <sup>a</sup>	Shapiro-Wilk			
	PortfolioType	Statistic	df	Sig.	Statistic	df	Sig.	
PortfolioReturns	GGL	.150	24	.172	.952	24	.303	
	NGL	.129	24	.200	.954	24	.332	

#### Tests of Normality

\*. This is a lower bound of the true significance.

a. Lilliefors Significance Correction

Table 5.10 shows that the results of the Shapiro Wilk test of normality for GGL and NGL for the 2008-2009 period show that there were no significant deviations from normality ( $W_{GGL} = .952$ , df = 24, p = .303), ( $W_{NGL} = .954$ , df = 24, p = .332).

As both p-values are higher than 0.05 the null hypothesis is not rejected and therefore given the lack of identified deviations in normality, an Independent Samples t-Test can be relied upon to test if there are significant differences between the monthly returns of GGL and NGL in the 2008-2009 sub-period.

#### 5.2.4.2 Test of Significance

The result of Levene's test for equality of Variance (Table 5.12) shows that there is insufficient evidence to reject the assumption of equal variances and so equal variances are assumed (**F** = .112, **p** = .739). The results of the Independent Samples t-Test (Table 5.12) show there is insufficient evidence to suggest that the monthly returns are different between the GGL portfolio (M=-.565, SD=5.77, n=24) and the NGL portfolio (M=-.938, SD=6.22, n=24) (Table 5.11) over the 2008-2009 period (t (46) = .215, p = .830). The research therefore, fails to reject the null hypothesis.

# Table 5.11: Groups Statistics – GGL versus NGL 2008-2009 period

Group Statistics

	PortfolioType	N	Mean	Std. Deviation	Std. Error Mean
PortfolioReturns	GGL	24	5646646	5.76918576	1.17763011
	NGL	24	9378675	6.22067129	1.26978921

# Table 5.12: Independent Samples t-Test – GGL versus NGL 2008-2009 period

	Independent Samples Test									
Levene's Test for Equality of Variances				t-test for Equality of Means						
					Mean	Std. Error	95% Confidence Differ	e Interval of the ence		
		F	Sig.	t	df	Sig. (2-tailed)	Difference	Difference	Lower	Upper
PortfolioReturns	Equal variances assumed	.112	.739	.215	46	.830	.37320292	1.73181330	-3.11275646	3.85916229
	Equal variances not assumed			.215	45.741	.830	.37320292	1.73181330	-3.11328858	3.85969441

# 5.2.5 Value / Large-Cap 3-Year Returns

Hypothesis 5, H0: There is no difference between the investment performance of a value/large-cap styled portfolio composed of green stocks and a value/large-cap styled portfolio composed of non-green stocks over the last 3 years?

The null hypothesis for this test is that there is no statistically significant difference between the tested groups.

# 5.2.5.1 Test of Normality

The sample data for the 3-year returns for the groups Green Value Large-Cap (GVL) and Non-Green Value Large-Cap (NVL) were analysed to test that each group was normally distributed. The null hypothesis is that the data sample is normally distributed and the alternative hypothesis is that the data is not normally distributed.

**Table 5.13**: Test of normality for Green and Non-Green Value/Large Cap Portfolios 3-Year

		Kolmogorov-Smirnov <sup>a</sup>			Shapiro-Wilk		
	PortfolioType	Statistic	df	Sig.	Statistic	df	Sig.
PortfolioReturns	GVL	.132	36	.116	.953	36	.133
	NVL	.150	36	.040	.967	36	.341

Tests of Normality

a. Lilliefors Significance Correction

Table 5.13 shows that the results of the Shapiro Wilk test of normality for GVL and NVL for the 3-Year period show that there were no significant deviations from normality  $(W_{GVL} = .953, df = 36, p = .113)$ ,  $(W_{NVL} = .967, df = 36, p = .341)$ .

As both p-values are higher than 0.05 the null hypothesis is not rejected and therefore given the lack of identified deviations in normality, an Independent Samples t-Test can be relied upon to test if there are significant differences between the monthly returns of GVL and NVL in the 3-Year sub-period.

# 5.2.5.2 Test of Significance

The result of Levene's test for equality of Variance (Table 5.15) shows that there is insufficient evidence to reject the assumption of equal variances and so equal variances are assumed (**F** = .011, **p** = .916). The results of the Independent Samples t-Test (Table 5.15) show there is insufficient evidence to suggest that the monthly returns are different between the GVL portfolio (**M**=1.02, **SD**=3.71, **n**=36) and the NVL portfolio (**M**=.96, **SD**=3.79, **n**=36) (Table 5.14) over a 3-Year period (**t** (70) = .071, **p** = .944). The research therefore, fails to reject the null hypothesis.

## Table 5.14: Groups Statistics – GVL versus NVL 3-Year period

Group Statistics

	PortfolioType	Ν	Mean	Std. Deviation	Std. Error Mean
PortfolioReturns	GVL	36	1.0229503	3.71420658	.61903443
	NVL	36	.9605736	3.79002822	.63167137

Table 5.15: Independent Samples t-Test – GVL versus NVL 3-Year period

	Independent Samples Test									
Levene's Test for Equality of Variances						t-test for Equality	of Means			
					Mean	Std. Error	95% Confidence Differ	e Interval of the ence		
		F	Sig.	t	df	Sig. (2-tailed)	Difference	Difference	Lower	Upper
PortfolioReturns	Equal variances assumed	.011	.916	.071	70	.944	.06237667	.88442769	-1.70155875	1.82631208
	Equal variances not assumed			.071	69.971	.944	.06237667	.88442769	-1.70157141	1.82632474

# 5.2.6 Value / Large-Cap 5-Year Returns

Hypothesis 6, H0: There is no difference between the investment performance of a value/large-cap styled portfolio composed of green stocks and a value/large-cap styled portfolio composed of non-green stocks over the last 5 years?

The null hypothesis for this test is that there is no statistically significant difference between the tested groups.

## 5.2.6.1 Test of Normality

The sample data for the 5-year returns for the groups Green Value Large-Cap (GVL) and Non-Green Value Large-Cap (NVL) were analysed to test that each group was normally distributed. The null hypothesis is that the data sample is normally distributed and the alternative hypothesis is that the data is not normally distributed.

 Table 5.16: Test of normality for Green and Non-Green Value/Large Cap Portfolios 5 

 Year

		Kolmogorov-Smirnov <sup>a</sup>			5	Shapiro-Wilk	
	PortfolioType	Statistic	df	Sig.	Statistic	df	Sig.
PortfolioReturns	GVL	.122	60	.027	.945	60	.009
	NVL	.105	60	.163	.979	60	.393

Tests of Normality

a. Lilliefors Significance Correction

Table 5.16 shows that the results of the Shapiro Wilk test of normality for GVL and NVL for the 5-Year period show that there were significant deviations from normality for GVL ( $W_{GVL} = .945$ , df = 60, p = .009), and no significant deviations from normality for NVL ( $W_{NVL} = .979$ , df = 60, p = .393).

As one of the group's p-value is less than 0.05 the null hypothesis is rejected and therefore as a deviation from normality is indicated, a Mann Whitney U-test is used to

test if there are significant differences between the monthly returns of GVL and NVL in the 5-Year sub-period.

# 5.2.6.2 Test of Significance

The results of the Mann Whitney U-test (Figures 5.17 and 5.18) show there is insufficient evidence to suggest that the monthly returns are different between the GVL portfolio with a mean rank score of **62.82** and the NVL portfolio with a mean rank score of **58.18** over a 5-Year period **(U=1661, p=.466).** The research therefore, fails to reject the null hypothesis.

Table 5.17: Mean Rank – GVL versus NVL 5-Year period

Nains						
	PortfolioType	Ν	Mean Rank	Sum of Ranks		
PortfolioReturns	GVL	60	62.82	3769.00		
	NVL	60	58.18	3491.00		
	Total	120				

Ranks

Table 5.18: Mann-Whitney U-test – GVL versus NVL 5-Year period

Test	Statisti	csa
		_

	PortfolioRetur ns
Mann-Whitney U	1661.000
Wilcoxon W	3491.000
Z	730
Asymp. Sig. (2-tailed)	.466

a. Grouping Variable: PortfolioType

# 5.2.7 Value / Large-Cap 10-Year Returns

Hypothesis 7, H0: There is no difference between the investment performance of a value/large-cap styled portfolio composed of green stocks and a value/large-cap styled portfolio composed of non-green stocks over the last 10 years?

The null hypothesis for this test is that there is no statistically significant difference between the tested groups.

## 5.2.7.1 Test of Normality

The sample data for the 10-year returns for the groups Green Value Large-Cap (GVL) and Non-Green Value Large-Cap (NVL) were analysed to test that each group was normally distributed. The null hypothesis is that the data sample is normally distributed and the alternative hypothesis is that the data is not normally distributed.

**Table 5.19**: Test of normality for Green and Non-Green Value/Large Cap Portfolios 10-Year

		Tes	ts of Norma	lity			
		Kolm	Kolmogorov-Smirnov <sup>a</sup> Shapiro-Wilk				
	PortfolioType	Statistic	df	Sig.	Statistic	df	Sig.
PortfolioReturns	GVL	.109	120	.001	.940	120	.000
	NVL	.069 120 .200" .977 120 .03					

\*. This is a lower bound of the true significance.

a. Lilliefors Significance Correction

Table 5.19 shows that the results of the Shapiro Wilk test of normality for GVL and NVL for the 10-Year period show that there were significant deviations from normality ( $W_{GVL}$  = .940, df = 120, p = .000), ( $W_{NGL}$  = .977, df = 120, p = .037).

As both p-values are less than 0.05 the null hypothesis is rejected and therefore as a deviation from normality is indicated, a Mann Whitney U-test is used to test if there are significant differences between the monthly returns of GGL and NGL in the 10-Year subperiod.

# 5.2.7.2 Test of Significance

The results of the Mann Whitney U-test (Figure 5.20 and 5.21) show there is insufficient evidence to suggest that the monthly returns are different between the GVL portfolio

with a mean rank score of **122.33** and the NGL portfolio with a mean rank score of **118.67** over a 10-Year period **(U=6980, p=.682).** The research therefore, fails to reject the null hypothesis.

Table 5.20: Mean Rank – GVL versus NVL 10-Year period

Ranks						
	PortfolioType	Ν	Mean Rank	Sum of Ranks		
PortfolioReturns	GVL	120	122.33	14680.00		
	NVL	120	118.67	14240.00		
	Total	240				

Table 5.21: Mann Whitney U test – GVL versus NVL 10-Year period

	PortfolioRetur
	ns
Mann-Whitney U	6980.000
Wilcoxon W	14240.000
Z	409
Asymp. Sig. (2-tailed)	.682

Test Statistics<sup>a</sup>

a. Grouping Variable: PortfolioType

# 5.2.8 Value / Large-Cap 2008-2009

Hypothesis 8, H0: There is no difference between the investment performance of a value/large-cap styled portfolio composed of green stocks and a value/large-cap styled portfolio composed of non-green stocks over the two-year crisis-period of 2008 and 2009?

The null hypothesis for this test is that there is no statistically significant difference between the tested groups.

#### 5.2.8.1 Test of Normality

The sample data for the 2008-2009 returns for the groups Green Value Large-Cap (GVL) and Non-Green Value Large-Cap (NVL) were analysed to test that each group was normally distributed. The null hypothesis is that the data sample is normally distributed and the alternative hypothesis is that the data is not normally distributed.

**Table 5.22**: Test of normality for Green and Non-Green Value/Large Cap Portfolios 2008-2009

		Kolmogorov-Smirnov <sup>a</sup>			Shapiro-Wilk			
	PortfolioType	Statistic	df	Sig.	Statistic	df	Sig.	
PortfolioReturns	GVL	.094	24	.200	.939	24	.155	
	NVL	.083	24	.200	.993	24	1.000	

Tests	of	Nor	ma	lity

\*. This is a lower bound of the true significance.

a. Lilliefors Significance Correction

Table 5.22 shows that the results of the Shapiro Wilk test of normality for GVL and NVL for the 2008-2009 period show that there were no significant deviations from normality ( $W_{GVL}$  = .939, df = 24, p = .155), ( $W_{NVL}$  = .993, df = 24, p = 1.000).

As both p-values are higher than 0.05 the null hypothesis is not rejected and therefore given the lack of identified deviations in normality, an Independent Samples t-Test can be relied upon to test if there are significant differences between the monthly returns of GVL and NVL in the 2008-2009 sub-period.

#### 5.2.8.2 Test of Significance

The result of Levene's test for equality of Variance (Table 5.24) shows that there is insufficient evidence to reject the assumption of equal variances and so equal variances are assumed (**F** = .004, **p** = .948). The results of the Independent Samples t-Test (Table 5.24) show there is insufficient evidence to suggest that the monthly returns are different between the GVL portfolio (**M**=-.308, **SD**=7.47, **n**=24) and the NGL portfolio (**M**=-.619, **SD**=7.08, **n**=24) (Table 5.23) over the 2008-2009 period (**t** (46) = .948, **p** = .883). The research therefore, fails to reject the null hypothesis.

# Table 5.23: Groups Statistics – GVL versus NVL 2008-2009 period

	Group Statistics									
	PortfolioType	Ν	Mean	Std. Deviation	Std. Error Mean					
PortfolioReturns	GVL	24	3078013	7.47437392	1.52570019					
	NVL	24	6191779	7.07729938	1.44464769					

# Table 5.24: Independent Samples t-Test - GVL versus NVL 2008-2009 period

Independent Samples Test										
Levene's Test for Equality of Variances										
					Mean	Std. Error	95% Confidence Differ	e Interval of the rence		
		F	Sig.	t	df	Sig. (2-tailed)	Difference	Difference	Lower	Upper
PortfolioReturns	Equal variances assumed	.004	.948	.148	46	.883	.31137667	2.10113493	-3.91798859	4.54074192
	Equal variances not assumed			.148	45.864	.883	.31137667	2.10113493	-3.91832804	4.54108137

#### 5.2.9 Growth / Mid-Cap 3-Year Returns

Hypothesis 9, H0: There is no difference between the investment performance of a growth/mid-cap styled portfolio composed of green stocks and a growth/mid-cap styled portfolio composed of non-green stocks over the last 3 years?

The null hypothesis for this test is that there is no statistically significant difference between the tested groups.

#### 5.2.9.1 Test of Normality

The sample data for the 3-year returns for the groups Green Growth Mid-Cap (GGM) and Non-Green Growth Mid-Cap (NGM) were analysed to test that each group was normally distributed. The null hypothesis is that the data sample is normally distributed and the alternative hypothesis is that the data is not normally distributed.

**Table 5.25**: Test of normality for Green and Non-Green Growth/Mid Cap Portfolios 3-Year

	Tests of Normany									
		Kolm	ogorov-Smir	nov <sup>a</sup>	Shapiro-Wilk					
	PortfolioType	Statistic	df	Sig.	Statistic	df	Sig.			
PortfolioReturns	GGM	.101	36	.200	.977	36	.640			
	NGM	.140	36	.073	.957	36	.179			

Toete of Normality

This is a lower bound of the true significance.

a. Lilliefors Significance Correction

Table 5.25 shows that the results of the Shapiro Wilk test of normality for GGM and NGM for the 3-Year period show that there were no significant deviations from normality ( $W_{GGM}$  = .977, df = 36, p = .640), ( $W_{NGL}$  = .957, df = 36, p = .179).

As both p-values are higher than 0.05 the null hypothesis is not rejected and therefore given the lack of identified deviations in normality, an Independent Samples t-Test can be relied upon to test if there are significant differences between the monthly returns of GGM and NGM in the 3-Year sub-period.

# 5.2.9.2 Test of Significance

The result of Levene's test for equality of Variance (Table 5.27) shows that there is insufficient evidence to reject the assumption of equal variances and so equal variances are assumed (**F** = **.139**, **p** = **.711**). The results of the Independent Samples t-Test (Table 5.27) show there is insufficient evidence to suggest that the monthly returns are different between the GGM portfolio (**M**=**.98**, **SD**=**3.56**, **n**=**36**) and the NGM portfolio (**M**=**.89**, **SD**=**3.28**, **n**=**36**) (Table 5.26) over a 3-Year period (**t** (**70**) = **.114**, **p** = **.711**). The research therefore, fails to reject the null hypothesis.

# Table 5.26: Groups Statistics – GGM versus NGM 3-Year period

	Group Statistics									
	PortfolioType	Ν	Mean	Std. Deviation	Std. Error Mean					
PortfolioReturns	GGM	36	.9787814	3.55581404	.59263567					
	NGM	36	.8867864	3.28062867	.54677145					

# Table 5.27: Independent Samples t-Test – GGM versus NGM 3-Year period

	Independent Samples Test										
Levene's Test for Equality of Variances t-test for Equality of Means											
F Sig t		df	Sig. (2-tailed)	Mean Difference	Std. Error Difference	95% Confidence Differ Lower	ence Upper				
PortfolioReturns	Equal variances assumed	.139	.711	.114	70	.909	.09199500	.80633495	-1.51618935	1.70017935	
	Equal variances not assumed			.114	69.551	.909	.09199500	.80633495	-1.51637203	1.70036203	

#### 5.2.10 Growth / Mid-Cap 5-Year Returns

Hypothesis 10, H0: There is no difference between the investment performance of a growth/mid-cap styled portfolio composed of green stocks and a growth/mid-cap styled portfolio composed of non-green stocks over the last 5 years?

The null hypothesis for this test is that there is no statistically significant difference between the tested groups.

#### 5.2.10.1 Test of Normality

The sample data for the 5-year returns for the groups Green Growth Mid-Cap (GGM) and Non-Green Growth Large-Cap (NGM) were analysed to test that each group was normally distributed. The null hypothesis is that the data sample is normally distributed and the alternative hypothesis is that the data is not normally distributed.

**Table 5.28**: Test of normality for Green and Non-Green Growth/Mid-Cap Portfolios 5-Year

reats of Normality									
		Kolm	ogorov-Smir	nov <sup>a</sup>	Shapiro-Wilk				
	PortfolioType	Statistic	df	Sig.	Statistic	df	Sig.		
PortfolioReturns	GGM	.084	60	.200	.978	60	.343		
	NGM	.105	60	.095	.969	60	.129		

Toete of Normality

This is a lower bound of the true significance.

a. Lilliefors Significance Correction

Table 5.28 shows that the results of the Shapiro Wilk test of normality for GGM and NGM for the 5-Year period show that there were no significant deviations from normality ( $W_{GGM}$  = .978, df = 60, p = .343), ( $W_{NGM}$  = .969, df = 60, p = .129).

As both p-values are higher than 0.05 the null hypothesis is not rejected and therefore given the lack of identified deviations in normality, an Independent Samples t-Test can be relied upon to test if there are significant differences between the monthly returns of GGM and NGM in the 5-Year sub-period.

# 5.2.10.2 Test of Significance

The result of Levene's test for equality of Variance (Table 5.30) shows that there is insufficient evidence to reject the assumption of equal variances and so equal variances are assumed (**F** = **.254**, **p** = **.615**). The results of the Independent Samples t-Test (Table 5.20) show there is insufficient evidence to suggest that the monthly returns are different between the GGM portfolio (**M**=**1.18**, **SD**=**3.14**, **n**=**60**) and the NGM portfolio (**M**=**.92**, **SD**=**2.87**, **n**=**60**) (Table 5.29) over a 5-Year period (**t** (**118**) = **.476**, **p** = **.635**). The research therefore fails to reject the null hypothesis.

# Table 5.29: Group Statistics – GGM versus NGM 5-Year period

Group Statistics								
	PortfolioType	Ν	Mean	Std. Deviation	Std. Error Mean			
PortfolioReturns	GGM	60	1.1826483	3.13990600	.40536012			
	NGM	60	.9211603	2.86597430	.36999569			

Course Chatiation

# Table 5.30: Independent Samples t-Test – GGM versus NGM 5-Year period

	Independent Samples Test											
Levene's Test for Equality of Variances t-test for E						t-test for Equality	or Equality of Means					
		F	Sig.	t	df	Sig. (2-tailed)	Mean Difference	Std. Error Difference	95% Confidence Differ Lower	e Interval of the ence Upper		
PortfolioReturns	Equal variances assumed	.254	.615	.476	118	.635	.26148800	.54882934	82534350	1.34831950		
	Equal variances not assumed			.476	117.030	.635	.26148800	.54882934	82543681	1.34841281		

## 5.2.11 Growth / Mid-Cap 10-Year Returns

Hypothesis 11, H0: There is no difference between the investment performance of a growth/mid-cap styled portfolio composed of green stocks and a growth/mid-cap styled portfolio composed of non-green stocks over the last 10 years?

The null hypothesis for this test is that there is no statistically significant difference between the tested groups.

#### 5.2.11.1 Test of Normality

The sample data for the 10-year returns for the groups Green Growth Mid-Cap (GGM) and Non-Green Growth Mid-Cap (NGM) were analysed to test that each group was normally distributed. The null hypothesis is that the data sample is normally distributed and the alternative hypothesis is that the data is not normally distributed.

**Table 5.31**: Test of normality for Green and Non-Green Growth/Mid Cap Portfolios 10-Year

reats of Normality									
		Kolm	ogorov-Smir	nov <sup>a</sup>	Shapiro-Wilk				
	PortfolioType	Statistic	df	Sig.	Statistic	df	Sig.		
PortfolioReturns	GGM	.084	120	.038	.948	120	.000		
	NGM	.102	120	.004	.947	120	.000		

Tests of Normality

a. Lilliefors Significance Correction

Table 5.31 shows that the results of the Shapiro Wilk test of normality for GGM and NGM for the 10-Year period show that there were significant deviations from normality ( $W_{GGM} = .948$ , df = 120, p = .000), ( $W_{NGM} = .947$ , df = 120, p = .000).

As both p-values are less than 0.05 the null hypothesis is rejected and therefore as a deviation from normality is indicated, a Mann Whitney U-test is used to test if there are significant differences between the monthly returns of GGM and NGM in the 10-Year sub-period.

#### 5.2.11.2 Test of Significance

The results of the Mann-Whitney U-test (Figures 5.32 and 5.33) show there is insufficient evidence to suggest that the monthly returns are different between the GGM portfolio with a mean rank score of **121.318** and the NGL portfolio with a mean rank score of **119.82** over a 10-Year period (**U=7118**, **p=.879**). The research therefore, fails to reject the null hypothesis.

Table 5.32: Mean Rank – GGM versus NGM 10-Year period

		Ranks		
	PortfolioType	N	Mean Rank	Sum of Ranks
PortfolioReturns	GGM	120	121.18	14542.00
	NGM	120	119.82	14378.00
	Total	240		

# Table 5.33: Mann Whitney U test – GGM versus NGM 10-Year period

Test Statistics <sup>a</sup>							
	PortfolioRetur ns						
Mann-Whitney U	7118.000						
Wilcoxon W	14378.000						
Z	152						
Asymp. Sig. (2-tailed)	.879						

a. Grouping Variable: PortfolioType

# 5.2.12 Growth / Mid-Cap 2008-2009

Hypothesis 12, H0: There is no difference between the investment performance of a growth/mid-cap styled portfolio composed of green stocks and a growth/mid-cap styled portfolio composed of non-green stocks over the two-year crisis-period of 2008 and 2009?

The null hypothesis for this test is that there is no statistically significant difference between the tested groups.

#### 5.2.12.1 Test of Normality

The sample data for the 2008-2009 returns for the groups Green Growth Mid-Cap (GGM) and Non-Green Growth Mid-Cap (NGM) were analysed to test that each group was normally distributed. The null hypothesis is that the data sample is normally distributed and the alternative hypothesis is that the data is not normally distributed.

**Table 5.34**: Test of normality for Green and Non-Green Growth/Mid Cap Portfolios2008-2009

		Kolmogorov-Smirnov <sup>a</sup>			Ś	Shapiro-Wilk	
	PortfolioType	Statistic	df	Sig.	Statistic	df	Sig.
PortfolioReturns	GGM	.153	24	.150	.973	24	.748
	NGM	.112	24	.200	.983	24	.939

Tests of Normality

\*. This is a lower bound of the true significance.

a. Lilliefors Significance Correction

Table 5.34 shows that the results of the Shapiro Wilk test of normality for GGM and NGM for the 2008-2009 period show that there were no significant deviations from normality ( $W_{GGM}$  = .973, df = 24, p = .748), ( $W_{NGM}$  = .983, df = 24, p = .939).

As both p-values are higher than 0.05 the null hypothesis is not rejected and therefore given the lack of identified deviations in normality, an Independent Samples t-Test can be relied upon to test if there are significant differences between the monthly returns of GGM and NGM in the 2008-2009 sub-period.

# 5.2.12.2 Test of Significance

The result of Levene's test for equality of Variance (Table 5.36) shows that there is insufficient evidence to reject the assumption of equal variances and so equal variances are assumed (**F** = .022, **p** = .883). The results of the Independent Samples t-Test (Table 5.36) show there is insufficient evidence to suggest that the monthly returns are different between the GGM portfolio (M=-.250, SD=7.67, n=24) and the NGM portfolio (M=-.337, SD=7.44, n=24) (Table 5.35) over the 2008-2009 period (t (46) = .040, p = .968). The research therefore, fails to reject the null hypothesis.

#### Table 5.35: Groups Statistics – GGM versus NGM 2008-2009 period

Group Statistics								
	PortfolioType	Ν	Mean	Std. Deviation	Std. Error Mean			
PortfolioReturns	GGM	24	2499742	7.67396261	1.56644106			
	NGM	24	3374079	7.44286524	1.51926851			

Table 5.36: Independent Samples t-Test – GGM versus NGM 2008-2009 period

Independent Samples Test											
Levene's Test for Equality of Variances					t-test for Equality of Means						
			Mean Std. Error Difference				e Interval of the ence				
		F	Sig.	t	df	Sig. (2-tailed)	Difference	Difference	Lower	Upper	
PortfolioReturns	Equal variances assumed	.022	.883	.040	46	.968	.08743375	2.18218111	-4.30506900	4.47993650	
	Equal variances not assumed			.040	45.957	.968	.08743375	2.18218111	-4.30517976	4.48004726	

54

# 5.2.13 Value / Mid-Cap 3-Year Returns

Hypothesis 13, H0: There is no difference between the investment performance of a value/mid-cap styled portfolio composed of green stocks and a value/mid-cap styled portfolio composed of non-green stocks over the last 3 years?

The null hypothesis for this test is that there is no statistically significant difference between the tested groups.

#### 5.2.13.1 Test of Normality

The sample data for the 3-year returns for the groups Green Value Mid-Cap (GVM) and Non-Green Value Mid-Cap (NVM) were analysed to test that each group was normally distributed. The null hypothesis is that the data sample is normally distributed and the alternative hypothesis is that the data is not normally distributed.

Table 5.37: Test of normality for Green and Non-Green Value/Mid Cap Portfolios 3-Year

Tests of Normality											
		Kolm	ogorov-Smir	nov <sup>a</sup>	Shapiro-Wilk						
	PortfolioType	Statistic	df	Sig.	Statistic	df	Sig.				
PortfolioReturns	GVM	.140	36	.071	.967	36	.344				
	NVM	.145	36	.052	.975	36	.566				

a. Lilliefors Significance Correction

Table 5.37 shows that the results of the Shapiro Wilk test of normality for GVM and NVM for the 3-Year period show that there were no significant deviations from normality ( $W_{GVM}$  = .967, df = 36, p = .344), ( $W_{NVL}$  = .975, df = 36, p = .566).

As both p-values are higher than 0.05 the null hypothesis is not rejected and therefore given the lack of identified deviations in normality, an Independent Samples t-Test can be relied upon to test if there are significant differences between the monthly returns of GVM and NVM in the 3-Year sub-period.

# 5.2.13.2 Test of Significance

The result of Levene's test for equality of Variance (Table 5.39) shows that there is insufficient evidence to reject the assumption of equal variances and so equal variances are assumed (**F** = **1.232**, **p** = **.271**). The results of the Independent Samples t-Test (Table 5.39) show there is insufficient evidence to suggest that the monthly returns are different between the GVM portfolio (**M**=**.98**, **SD**=**3.68**, **n**=**36**) and the NVM portfolio (**M**=**1.0**, **SD**=**4.28**, **n**=**36**) (Table 5.38) over a 3-Year period (**t** (**70**) = **-0.23**, **p** = **.982**). The research therefore, fails to reject the null hypothesis.

# Table 5.38: Groups Statistics – GVM versus NVM 3-Year period

	PortfolioType	Ν	Mean	Std. Deviation	Std. Error Mean				
PortfolioReturns	GVM	36	.9795714	3.67656786	.61276131				
	NVM	36	1.0009653	4.27778782	.71296464				

Crown Statistics

# Table 5.39: Independent Samples t-Test – GVM versus NVM 3-Year period

	Independent Samples Test											
Levene's Test for Equality of Variances				test for Equality of Means								
							Mean	Std. Error	95% Confidence Interval of the Std. Error Difference			
		F	Sig.	t	df	Sig. (2-tailed)	Difference	Difference	Lower	Upper		
PortfolioReturns	Equal variances assumed	1.232	.271	023	70	.982	02139389	.94010372	-1.89637163	1.85358385		
	Equal variances not assumed			023	68.453	.982	02139389	.94010372	-1.89711666	1.85432889		

#### 5.2.14 Value / Mid-Cap 5-Year Returns

Hypothesis 14, H0: There is no difference between the investment performance of a value/mid-cap styled portfolio composed of green stocks and a value/mid-cap styled portfolio composed of non-green stocks over the last 5 years?

The null hypothesis for this test is that there is no statistically significant difference between the tested groups.

#### 5.2.14.1 Test of Normality

The sample data for the 5-year returns for the groups Green Value Mid-Cap (GVM) and Non-Green Value Large-Cap (NVM) were analysed to test that each group was normally distributed. The null hypothesis is that the data sample is normally distributed and the alternative hypothesis is that the data is not normally distributed.

**Table 5.40**: Test of normality for Green and Non-Green Value/Mid-Cap Portfolios 5-Year

		Kolmogorov-Smirnov <sup>a</sup>			Ś	Shapiro-Wilk	
	PortfolioType	Statistic	df	Sig.	Statistic	df	Sig.
PortfolioReturns	GVM	.117	60	.039	.970	60	.148
	NVM	.100	60	.200	.982	60	.533

Tests of Normality

\*. This is a lower bound of the true significance.

a. Lilliefors Significance Correction

Table 5.40 shows that the results of the Shapiro Wilk test of normality for GVM and NVM for the 5-Year period show that there were no significant deviations from normality  $(W_{GVM} = .97o, df = 60, p = .148), (W_{NVM} = .982, df = 60, p = .533).$ 

As both p-values are higher than 0.05 the null hypothesis is not rejected and therefore given the lack of identified deviations in normality, an Independent Samples t-Test can be relied upon to test if there are significant differences between the monthly returns of GVM and NVM in the 5-Year sub-period.

# 5.2.14.2 Test of Significance

The result of Levene's test for equality of Variance (Table 5.42) shows that there is insufficient evidence to reject the assumption of equal variances and so equal variances are assumed (**F** = **.957**, **p** = **.330**). The results of the Independent Samples t-Test (Table 5.42) show there is insufficient evidence to suggest that the monthly returns are different between the GVM portfolio (**M**=**1.12**, **SD**=**3.44**, **n**=**60**) and the NVM portfolio (**M**=**.99**, **SD**=**3.76**, **n**=**60**) (Table 5.41) over a 5-Year period (**t** (**118**) = **.186**, **p** = **.853**). The research therefore, fails to reject the null hypothesis.
### Table 5.41: Group Statistics – GVM versus NVM 5-Year period

_	Group Statistics								
		PortfolioType	Ν	Mean	Std. Deviation	Std. Error Mean			
Γ	PortfolioReturns	G∨M	60	1.1181453	3.44127625	.44426685			
		NVM	60	.9956633	3.76365440	.48588569			

### Table 5.42: Independent Samples t-Test - GVM versus NVM 5-Year period

	Independent Samples Test									
Levene's Test for Equality of Variances t-test for Equality of Means										
Mean Mean			Std. Error	95% Confidence Differ	e Interval of the ence					
PortfolioReturns	Equal variances		oig.		u	org. (2-taneu)	Difference	Difference	201101	oppor
1 officient citating	assumed	.957	.330	.186	118	.853	.12248200	.65837523	-1.18128020	1.42624420
	Equal variances not assumed			.186	117.066	.853	.12248200	.65837523	-1.18138793	1.42635193

### 5.2.15 Value / Mid-Cap 10-Year Returns

Hypothesis 15, H0: There is no difference between the investment performance of a value/mid-cap styled portfolio composed of green stocks and a value/mid-cap styled portfolio composed of non-green stocks over the last 10 years?

The null hypothesis for this test is that there is no statistically significant difference between the tested groups.

#### 5.2.15.1 Test of Normality

The sample data for the 10-year returns for the groups Green Value Mid-Cap (GVM) and Non-Green Value Mid-Cap (NVM) were analysed to test that each group was normally distributed. The null hypothesis is that the data sample is normally distributed and the alternative hypothesis is that the data is not normally distributed. **Table 5.43**: Test of normality for Green and Non-Green Value/Mid Cap Portfolios 10-Year

Toolo of Normany									
		Kolmogorov-Smirnov <sup>a</sup>			Shapiro-Wilk				
	PortfolioType	Statistic	df	Sig.	Statistic	df	Sig.		
PortfolioReturns	GVM	.121	120	.000	.923	120	.000		
	NVM	.060	120	.200	.949	120	.000		

Tests of Normality

This is a lower bound of the true significance.

a. Lilliefors Significance Correction

Table 5.43 shows that the results of the Shapiro Wilk test of normality for GVM and NVM for the 10-Year period show that there were significant deviations from normality ( $W_{GVM}$  = .923, df = 120, p = .000), ( $W_{NVM}$  = .949, df = 120, p = .000).

As both p-values are less than 0.05 the null hypothesis is rejected and therefore as a deviation from normality is indicated, a Mann Whitney U-Test is used to test if there are significant differences between the monthly returns of GVM and NVM in the 10-Year sub-period.

### 5.2.15.2 Test of Significance

The results of the Mann-Whitney U-test (Figures 5.44 and 5.45) show there is insufficient evidence to suggest that the monthly returns are different between the GVM portfolio with a mean rank score of **118.93** and the NVL portfolio with a mean rank score of **112.07** over a 10-Year period (**U=7012**, **p=.727**). The research therefore, fails to reject the null hypothesis.

Ranks								
	PortfolioType	Ν	Mean Rank	Sum of Ranks				
PortfolioReturns	GVM	120	118.93	14272.00				
	NVM	120	122.07	14648.00				
	Total	240						

Table 5.44: Mean Rank – GVM versus NVM 10-Year period

### Table 5.45: Mann Whitney U Test – GVM versus NVM 10-Year period

Test Statistics					
	PortfolioRetur ns				
Mann-Whitney U	7012.000				
Wilcoxon W	14272.000				
Z	350				
Asymp. Sig. (2-tailed)	.727				

Tant Ctationian 3

a. Grouping Variable: PortfolioType

### 5.2.16 Value / Mid-Cap 2008-2009

Hypothesis 16, H0: There is no difference between the investment performance of a value/mid-cap styled portfolio composed of green stocks and a value/mid-cap styled portfolio composed of non-green stocks over the two-year crisis-period of 2008 and 2009?

The null hypothesis for this test is that there is no statistically significant difference between the tested groups.

#### 5.2.16.1 Test of Normality

The sample data for the 2008-2009 returns for the groups Green Value Mid-Cap (GVM) and Non-Green Value Mid-Cap (NVM) were analysed to test that each group was normally distributed. The null hypothesis is that the data sample is normally distributed and the alternative hypothesis is that the data is not normally distributed.

**Table 5.46**: Test of normality for Green and Non-Green Value/Mid Cap Portfolios 2008-2009

		Kolm	ogorov-Smir	nov <sup>a</sup>	Shapiro-Wilk		
	PortfolioType	Statistic	df	Sig.	Statistic	df	Sig.
PortfolioReturns	GVM	.104	24	.200	.937	24	.138
	NVM	.106	24	.200	.956	24	.364

**Tests of Normality** 

\*. This is a lower bound of the true significance.

a. Lilliefors Significance Correction

Table 5.46 shows that the results of the Shapiro Wilk test of normality for GVM and NVM for the 2008-2009 period show that there were no significant deviations from normality  $(W_{GVM} = .937, df = 24, p = .138)$ ,  $(W_{NVM} = .983, df = 24, p = .364)$ .

As both p-values are higher than 0.05 the null hypothesis is not rejected and therefore given the lack of identified deviations in normality, an Independent Samples t-Test can be relied upon to test if there are significant differences between the monthly returns of GVM and NVM in the 2008-2009 sub-period.

### 5.2.16.2 Test of Significance

The result of Levene's test for equality of Variance (Table 5.48) shows that there is insufficient evidence to reject the assumption of equal variances and so equal variances are assumed (**F** = .046, **p** = .831). The results of the Independent Samples t-Test (Table 5.48) show there is insufficient evidence to suggest that the monthly returns are different between the GVM portfolio (**M**=.291, **SD**=8.72, **n**=24) and the NVM portfolio (**M**=.867, **SD**=8.76, **n**=24) (Table 5.47) over the 2008-2009 period (**t** (46) = -.228, **p** = .821). The research therefore, fails to reject the null hypothesis.

### Table 5.47: Groups Statistics – GVM versus NVM 2008-2009 period

Group	Statistics
-------	------------

	PortfolioType	Ν	Mean	Std. Deviation	Std. Error Mean
PortfolioReturns	G∨M	24	.2913700	8.71500193	1.77894232
	NVM	24	.8666425	8.75764365	1.78764652

### Table 5.48: Independent Samples t-Test – GVM versus NVM 2008-2009 period

	Independent Samples Test									
	Levene's Test for Equality of Variances t-test for Equality of Means									
							Mean	Std. Error	95% Confidence Differ	e interval of the ence
		F	Sig.	t	df	Sig. (2-tailed)	Difference	Difference	Lower	Upper
PortfolioReturns	Equal variances assumed	.046	.831	228	46	.821	57527250	2.52196667	-5.65172811	4.50118311
	Equal variances not assumed			228	45.999	.821	57527250	2.52196667	-5.65173137	4.50118637

### 5.3 Time-Weighted Returns

This section looks at the results of 3-year, 5-year and 10-year performance returns of green and non-green portfolios across the four styles of growth/large-cap, value/large-cap, growth/mid-cap and value/mid-cap. The time-weighted returns were calculated by chain-linking the monthly returns generated by Eikon to produce 3-year, 5-year, 10-year returns and 2008-2009. The returns were then annualised for the purpose of comparison.

#### 5.3.1 Growth / Large-Cap

**Figure 5.1**: Geometric Returns: Green versus Non-Green for Growth/Large-Cap portfolios



GGL= Green Growth Large-Cap, NGL= Non-Green Growth Large-Cap

Figure 5.1 shows the 3-Year, 5-Year, 10-Year and 2008-2009 returns of Green Growth Large-Cap and Non-Green Growth Large-Cap portfolios. The green portfolio outperforms the non-green portfolio in all four sub periods. The 3-Year annualised return for the green portfolio was 12.93%, compared to 10.06% for the non-green portfolio. The 5-Year annualised return for the green portfolio was 14.81%, compared

to 11.33% for the non-green portfolio. The 10-Year annualised return for the green portfolio was 8.43%, compared to 6.41% for the non-green portfolio. The 2008-2009 annualised return for the green portfolio was -8.38%, compared to -12.72% for the non-green portfolio. The highest return for growth/large-cap portfolios was the green portfolio's 5-year return of 14.81%. The lowest return for growth/large-cap portfolios was the non-green portfolio's 2008-2009 return of -12.72%.

### 5.3.2 Value / Large-Cap



Figure 5.2: Geometric Returns: Green versus Non-Green for Value/Large-Cap portfolios

GVL= Green Value Large-Cap, NVL= Non-Green Value Large-Cap

Figure 5.2 shows the 3-Year, 5-Year, 10-Year and 2008-2009 returns of Green Value Large-Cap and Non-Green Value Large-Cap portfolios. The green portfolio outperforms the non-green portfolio in all three sub periods. The 3-Year annualised return for the green portfolio was 12.09%, compared to 11.22% for the non-green portfolio. The 5-Year annualised return for the green portfolio was 15.68%, compared to 11.22% for the non-green portfolio. The 10-Year annualised return for the green portfolio was 6.71%,

compared to 4.81% for the non-green portfolio. The 2008-2009 annualised return for the green portfolio was -6.6%, compared to -9.85% for the non-green portfolio. The highest return for value/large-cap portfolios was the green portfolio's 5-year return of 15.68%. The lowest return for value/large-cap portfolios was the non-green portfolio's 2008-2009 return of -9.85%.

### 5.3.3 Growth / Mid-Cap



Figure 5.3: Geometric Returns: Green versus Non-Green for Growth/Mid-Cap portfolios

GGM= Green Growth Mid-Cap, NGM= Non-Green Growth Mid-Cap

Figure 5.3 shows the 3-Year, 5-Year, 10-Year returns and 2008-2009 of Green Growth Mid-Cap and Non-Green Growth Mid-Cap portfolios. The green portfolio outperforms the non-green portfolio in all three sub periods. The 3-Year annualised return for the green portfolio was 11.59%, compared to 10.48% for the non-green portfolio. The 5-Year annualised return for the green portfolio was 14.49%, compared to 11.09% for the non-green portfolio. The 10-Year annualised return for the green portfolio was 7.77%, compared to 7.45% for the non-green portfolio. The 2008-2009 annualised return for

the green portfolio was -6.29%, compared to -7.06% for the non-green portfolio. The highest return for growth/mid-cap portfolios was the green portfolio's 5-year return of 14.49%. The lowest return for growth/mid-cap portfolios was the non-green portfolio's 10-year return of -7.06%.

### 5.3.4 Value / Mid-Cap



Figure 5.4: Geometric Returns: Green versus Non-Green for Value/Mid-Cap portfolios

GVM= Green Value Mid-Cap, NVM= Non-Green Value Mid-Cap

Figure 5.4 shows the 3-Year, 5-Year, 10-Year and 2008-2009 returns of Green Value Mid-Cap and Non-Green Value Mid-Cap portfolios. The green portfolio outperforms the non-green portfolio in the 3-Year and 5-year period. The non-green portfolio outperforms the green portfolio in the 10-Year period. The 3-Year annualised return for the green portfolio was 11.54%, compared to 11.51% for the non-green portfolio. The 5-Year annualised return for the green portfolio was 13.49%, compared to 11.70% for the non-green portfolio. The 10-Year annualised return for the green portfolio. The 10-Year annualised return for the green portfolio. The 3.49%, compared to 11.70% for the non-green portfolio. The 10-Year annualised return for the green portfolio. The 3.49%, compared to 11.70% for the non-green portfolio.

the green portfolio was -0.67%, compared to 6.32% for the non-green portfolio. The highest return for value/mid-cap portfolios was the green portfolio's 5-year return of 13.49%. The lowest return for value/mid-cap portfolios was the green portfolio's 2008-2009 return of -0.67%.

### 5.4 Sharpe Ratio

This section looks at the results of 3-year, 5-year, 10-year and 2008-2009 period's Sharpe ratio of green and non-green portfolios across the four styles of large/growth, large value, mid/growth and mid/value.

### 5.4.1 Growth / Large-Cap



Figure 5.5: Sharpe ratio - Green versus Non-Green for Growth/Large-Cap portfolios

GGL= Green Growth Large-Cap, NGL= Non-Green Growth Large-Cap

Figure 5.47 shows the 3-Year, 5-Year, 10-Year and 2008-2009 Sharpe ratio of Green Growth Large-Cap and Non-Green Growth Large-Cap portfolios. The green portfolio show a better Sharpe ratio than the non-green portfolios in all three sub periods. The 3-Year Sharpe ratio for the green portfolio was 1.14, compared to .93 for the non-green portfolio. The 5-Year Sharpe ratio for the green portfolio for the green portfolio was 1.48, compared to 1.15 for the non-green portfolio. The 10-Year Sharpe ratio for the green portfolio was .56

compared to .38 for the non-green portfolio. The 2008-2009 Sharpe ratio for the green portfolio was -0.58 compared to -0.75 for the non-green portfolio. The highest Sharpe ratio for growth/large-cap portfolios was the green portfolio's 5-year ratio of 1.48. The lowest Sharpe ratio for growth/large-cap portfolios was the non-green portfolio's 2008-2009 ratio of -0.75.

### 5.4.2 Value / Large-Cap



Figure 5.6: Sharpe ratio - Green versus Non-Green for Value/Large-Cap portfolios

GVL= Green Value Large-Cap, NGL= Non-Green Value Large-Cap

Figure 5.8 shows the 3-Year, 5-Year, 10-Year and 2008-2009 Sharpe ratio of Green Value Large-Cap and Non-Green Value Large-Cap portfolios. The green portfolio show a better Sharpe ratio than the non-green portfolios in all three sub periods. The 3-Year Sharpe ratio for the green portfolio was 0.96, compared to .87 for the non-green portfolio. The 5-Year Sharpe ratio for the green portfolio was 1.36, compared to 1.05 for the non-green portfolio. The portfolio. The 10-Year Sharpe ratio for the green portfolio for the green portfolio was .35 compared to .23 for the non-green portfolio. The 2008-2009 Sharpe ratio for the green portfolio was -0.38

compared to -0.54 for the non-green portfolio. The highest Sharpe ratio for growth/large-cap portfolios was the green portfolio's 5-year ratio of 1.36. The lowest Sharpe ratio for growth/large-cap portfolios was the non-green portfolio's 2008-2009 ratio of -0.54.

### 5.4.3 Growth / Mid-Cap



Figure 5.7: Sharpe ratio - Green versus Non-Green for Growth/Mid-Cap portfolios

GGM= Green Growth Mid-Cap, NGM= Non-Green Growth Mid-Cap

Figure 5.9 shows the 3-Year, 5-Year, 10-Year and 2008-2009 Sharpe ratio of Green Growth Mid-Cap and Non-Green Growth Mid-Cap portfolios. The green portfolio show a better Sharpe ratio than the non-green portfolios in all three sub periods. The 3-Year Sharpe ratio for the green portfolio was 0.96, compared to .94 for the non-green portfolio. The 5-Year Sharpe ratio for the green portfolio was 1.33, compared to 1.12 for the non-green portfolio. The 10-Year Sharpe ratio for the green portfolio was .42 compared to .41 for the non-green portfolio. The 2008-2009 Sharpe ratio for the green portfolio was -0.36 compared to -0.4 for the non-green portfolio. The highest Sharpe

ratio for growth/mid-cap portfolios was the green portfolio's 5-year ratio of 1.33. The lowest Sharpe ratio for growth/mid-cap portfolios was the non-green portfolio's 2008-2009 of -0.4.

### 5.4.4 Value / Mid-Cap

Figure 5.8: Sharpe ratio - Green versus Non-Green for Value/Mid-Cap portfolios



GVM= Green Value Mid-Cap, NGM= Non-Green Value Mid-Cap

Figure 5.10 shows the 3-Year, 5-Year, 10-Year and 2008-2009 Sharpe ratio of Green Value Mid-Cap and Non-Green Value Mid-Cap portfolios. The green portfolio show a better Sharpe ratio than the non-green portfolios in the 3-Year and 5-Year periods. The non-green portfolio shows a better Sharpe ratio than the green portfolios in the 10-Year period. The 3-Year Sharpe ratio for the green portfolio was 0.92, compared to .79 for the non-green portfolio. The 5-Year Sharpe ratio for the green portfolio was 1.13, compared to 0.9 for the non-green portfolio. The 10-Year Sharpe ratio for the green portfolio. The 2008-2009 Sharpe ratio for the green portfolio. The green portfolio. The 10-Year Sharpe ratio for the green portfolio. The green portfolio. The 10-Year Sharpe ratio for the green portfolio. The green portfolio. The 10-Year Sharpe ratio for the green portfolio. The green portfolio. The 10-Year Sharpe ratio for the green portfolio. The green portfolio. The 10-Year Sharpe ratio for the green portfolio. The green portfolio. The 10-Year Sharpe ratio for the green portfolio. The green portfolio. The 2008-2009 Sharpe ratio for the green portfolio was -0.12 compared to .11 for the non-green portfolio. The

highest Sharpe ratio for growth/mid-cap portfolios was the green portfolio's 5-year ratio of 1.13. The lowest Sharpe ratio for growth/mid-cap portfolios was the green portfolio's 2008-2009 ratio of -0.12.

## 6 Discussion

In this chapter, the findings of the statistical analysis for differences between green and non-green portfolios in the four investment styles of growth/large-cap, growth/mid-cap, value/large-cap and value/mid-cap for the 3-year, 5-year and 10-year periods along with the 2008-2009 crisis period will be discussed. Additionally, the time-weighted geometric returns for the four sub-periods will also be discussed. The Sharpe ratio, indicating risk adjusted returns, will also be looked at in order to identify if any difference in portfolio performance between portfolios is due to risk. Finally, limitations of the research will be discussed.

From the literature, most studies have reported no significant difference between SRI performance and traditional investing (Junkus & Berry, 2014). This study however focuses on the behaviour of environmental stocks to discover if there is a significant difference in the investment performance of green and non-green stocks. The results from this test suggest that there are no statistically significant differences between the performance of green stocks versus non-green stocks over all four investment styles of growth/large cap, growth mid-cap, value large-cap, value mid-cap for the 3-year, 5-year and 10-year periods and the crisis-period 2008-2009. The findings therefore suggest alignment with Reveilli & Viviani (2015) meta-analysis of 85 previous studies, in that as with SRI investments, there is no real cost or benefit to investing in green stocks.

Although the results indicate no significant difference in the performance of green stocks and non-green stocks across the four investments styles and sub-periods, the results of the geometric time-weighted returns show the investment performance of portfolios composed of green stocks outperformed those composed of non-green stocks in all investment styles and sub-periods with the exception of the 10-Year and 2008-2009 returns for value/mid-cap. These findings are in contrast with Muñoz et al. (2014) study on green funds who found that green underperformed in normal periods, however their study was comparing green funds with SRI funds as opposed to green

stocks versus non-green stocks. The research is aligned with the results from Lesser et al. (2014) who found green funds out performed SRI funds in non-crisis period, again however, their study was comparing green funds with SRI funds as opposed to green stocks versus non-green stocks. However, in contrast to Lesser et. al (2014) who showed that green funds underperformed in crisis-period, this study shows that green stocks underperformed in crisis-period in only one of the investment styles, value/mid-cap.

The geometric returns as outlined in the findings in the previous chapter are organised per investment style detailing performance for green versus non-green for the four different sub-periods. However, this section will discuss the cross-sectional returns for all investment styles per sub-period to determine best and worst investment styles per period. The 3-Year period's best return was the GGL portfolio with 12.93% and the worst performance for 3-Year period was the NGL portfolio return of 10.06%. The 5-Year period's best return was the GVL portfolio return of 15.68%, whereas the worst performance for the 5-Year period was the NVM portfolio return of 11.09%. The 10-Year period's best return was the NVM portfolio return of 9.03% and the worst performance for the 10-Year period was the NVL portfolio return of 4.81%. The 2008-2009 period's best return was the NVM portfolio return of 6.32% and the worst performance for the 2008-2009 period was the NGL portfolio return of -12.72%. The returns indicate that green large-cap portfolios in both value and growth were the best performers in the 3-Year and 5-Year periods whereas the non-green value mid-cap was the best performer in the 10-Year and 2008-2009 period. The non-green value/mid-cap portfolio was the only portfolio to show positive returns in the 2008-2009 period with 6.32% and the nongreen value/mid-cap portfolio with -0.67% meant that both green and non-green value/mid-cap portfolios outperformed all other styles in the 2008-2009 period with the nearest best performer being the green growth/mid-cap portfolio with -6.29%.

In the non-crisis periods therefore, this study shows that the green large-cap portfolios were the best performers, and in the crisis period the value-mid cap portfolios were the

best performers with the non-green value-mid cap being the only portfolio to have positive returns for the 2008-2009 period.

As the Sharpe ratio is a measurement of the portfolios excess return per unit of risk as defined by the portfolios standard deviation, its use will enable a better comparison of portfolio performance on a risk-adjusted basis (Schröder, 2007; Statman & Glushkov, 2016). As can be seen from Figures 5.5 to 5.8 in the Findings section, Sharpe ratios are consistent with the geometric time-weighted returns in that they indicate that after returns are adjusted for risk, the investment performance of portfolios composed of green stocks outperform those composed of non-green stocks in all investment styles and sub-periods with the exception of the 10-Year and 2008-2009 returns for value/mid-cap. This indicates that whether a green portfolio outperforms or underperforms its non-green equivalent in the findings above is not due to the element of risk.

However, a closer look at the geometric returns and the Sharpe ratio indicate that the amount by which a portfolio outperforms or underperforms is affected by the element of risk. This is evident in the 2008-2009 period where standard deviation of returns for all portfolios is high, with green value/mid-cap and non-green value/mid-cap being the highest at 8.72 and 8.76 respectively (see Appendix A for statistical descriptives for all portfolios). This would suggest that the level of outperformance would be reduced once returns are adjusted for risk. As can be seen from comparing the 2008-2009 geometric returns with 2008-2009 Sharpe ratios in Figures 5.4 and 5.8, the level of outperformance of the non-green portfolio over the green portfolio is reduced once the returns are adjusted by risk. The Sharpe ratio of .11 for the non-green value/mid-cap portfolio however, is still the only portfolio for the 2008-2009 period to be positive even after returns are risk-adjusted. The negative Sharpe ratio for all other portfolios in the 2008-2009 period indicates that investment performance was worse for these portfolios than investing in the risk-free rate alone.

## 7 Conclusion

The main research objective was to construct and analyse portfolios comprised of green and non-green stocks based on the MSCI ESG Environmental pillar scores and categorised within appropriate investment styles with the objective of comparing their performance with each other over 3-Year, 5-Year and 10-Year periods, and over the 2-Year crisis-period 2008-2009. The drivers for this research were the gaps in the literature based on studies analysing performance of funds rather than the performance of stocks in the market and on the existing literature's focus on SRI and ESG rather than green stocks. The importance of the research is to discover if investing solely in green stocks comes at a cost to investment performance returns and also to identify any trends in the performance returns of green stocks within a cross-sectional analysis across four investment styles, which may be informative and helpful for investors when choosing stocks to invest in.

The research findings indicate no statistically significant difference between the performance of green stocks and non-green stocks across the four investment styles of growth/large-cap, value/large-cap, growth/mid-cap and value/mid-cap within the 3-Year, 5-Year, 10-Year and 2008-2009 periods. These results suggest no benefit or cost to investors who wish to invest solely in green stocks. The implication of this is that investors who wish to positively allocate stocks with better environmental credentials than their peers, would not be penalised in performance terms for so doing.

The research also revealed that the time-weighted returns of green stocks outperformed non-green stocks in growth/large-cap, value/large-cap, growth mid-cap, investment styles over 3-Year, 5-Year, 10-Year and 2008-2009 periods. The timeweighted returns of green stocks also outperformed non-green stocks in the value/midcap style over the 3-Year and 5-year periods. However, the time-weighted returns of non-green stocks outperformed the green stocks in the value/mid-cap style over the 10-Year and 2008-2009 periods. The research also demonstrated that the risk-adjusted returns were consistent with the time-weighted returns. Sharpe ratios showed that risk-adjusted returns of green stocks out-performed non-green stocks in growth/large-cap, growth mid-cap, value/large-cap, investment styles over 3-Year, 5-Year, 10-Year and 2008-2009 periods. Sharpe ratios also showed that risk-adjusted returns of green stocks outperformed non-green stocks in the value/mid-cap style over the 3-Year and 5-year periods. However, the Sharpe ratios showed that risk-adjusted returns of non-green stocks outperformed the green stocks in the value/mid-cap style over the 10-Year and 2008-2009 periods.

Limitations of the research include the exclusion of small-cap stocks from the research. The research used the constituents of the MSCI ACWI which includes large-cap and midcap stocks but not small-cap stocks. Another limitation of the research was that it covered only one crisis-period, the 2008 financial crisis. It is not possible therefore to draw any conclusions regarding the possible existence of trends of investment performance of green and non-green stocks post crisis periods.

Some obvious recommendations therefore emerge from the limitations of the research. A more comprehensive analysis could be achieved if similar research was carried out which included the constituents of an index containing small-cap stocks, such as MSCI ACWI Small Cap Index. Additionally, further research which covered more than one crisis-period would allow for deeper analysis. Finally, cut-off points of top 30% and bottom 30% were used to determine green and non-green stocks, various different cutoff points could be used to investigate whether there was an optimal point where environmental ratings positively affected investment performance.

75

## References

Bauer, R., Denva, J. & Otten, R. (2006), "The Ethical Mutual Fund Performance Debate: New Evidence from Canada", *Journal of Business Ethics*, vol. 70, pp 111-124.

Bauman, W.S., Conover, M.C. & Miller, R.E. (1998), "Growth versus value and large-cap versus small-cap stocks in international markets", *Financial Analysts Journal*, vol. 54, no. 2, pp. 75-89.

Berry, T. & Junkus, J. (2013), "Socially Responsible Investing: An Investor Perspective", *Journal of Business Ethics*, vol. 112, no. 4, pp. 707-720.

Boerner, H. (2007), "Your Company's ESG-Environmental, Social, and Governance Factors are Mattering More Now to Institutional Investors", *Corporate Finance Review*, vol. 12, no. 2, pp. 40-43.

Borys, M. & Zemčik, P. (2011), "Size and Value Effects in the Visegrad Countries", *Emerging Markets Finance & Trade*, vol. 47, no. 3, pp. 50-68.

Boulatoff, C. & Boyer, C.M. (2009), "Green Recovery: How Are Environmental Stocks Doing?", *The Journal of Wealth Management*, vol. 12, no. 2, pp. 9-20.

Bryman, A. & Bell, E. (2011), *Business Research Methods*, 3<sup>rd</sup> Ed., Oxford University Press.

Brzeszczyński, J. & McIntosh, G. (2014), "Performance of Portfolios Composed of British SRI Stocks", *Journal of Business Ethics*, vol. 120 no.3, pp. 335-362.

Capelle-Blancard, G, & Monjon, S (2014), "The Performance of Socially Responsible Funds: Does the Screening Process Matter?", *European Financial Management*, vol. 20, no.3, pp. 494-52.

Capual, C., Rowley, I. & Sharpe, W.F. (1993), "International value and growth stock returns", *Financial Analysts Journal*, vol. 49, no. 1, pp. 27.

Carlsson-Sweeny, M. (2014), "ESG - Busting the myths", *Governance Directions*, vol. 66, no. 4, pp. 220-222.

Chan, L., Hsiu-Lang, C. & Lakonishok, J. (2002), "On Mutual Fund Investment Styles", *Review of Financial Studies*, vol. 15, no. 5, pp. 1407-1437.

Chang, C.E., Nelson, W.A. & Witte, H.D. (2012), "Do green mutual funds perform well?", *Management Research Review*, vol. 35, no. 8, pp. 693-708.

Chang, C.E. & Witte, H.D. (2010), "Performance Evaluation of U.S. Socially Responsible Mutual Funds: Revisiting Doing Good and Doing Well", *American Journal of Business*, vol. 25, no. 1, pp. 9-21.

Chia, C.P., Goldberg, L.R., Owyong, D.T., Shepard, P. & Stoyanov, T. (2009), "Is There a Green Factor?", *Journal of Portfolio Management*, vol. 35, no. 3, pp. 34-40.

Climent, F. & Soriano, P. (2011), "Green and Good? The Investment Performance of US Environmental Mutual Funds", *Journal of Business Ethics*, vol. 103, no. 2, pp. 275-287.

Dolinar, D. (2013), "Test of the Fama-French Three-Factor Model in Croatia", *UTMS Journal of Economics*, vol. 4, no. 2, pp. 101-112.

Dor, A., Budinger, V., Dynkin, L. & Leech, K. (2008), "Constructing Peer Benchmarks for Mutual Funds: A Style Analysis-Based Approach", *Journal of Portfolio Management*, vol. 34, no. 2, pp. 65-77.

Eisinberg, L.C. (2000), "Small-cap vs. large-cap: What's the story?", *Agency Sales*, vol. 30, no. 12, pp. 32.

Enterprise Ireland (2006), *Profiles of Irish Software and Service Companies Supplying the Financial Sector* [ONLINE] Available at: <u>http://www.enterprise-</u> <u>ireland.or.jp/uploads/fin\_services\_directory06%20Complete.pdf</u> [Accessed 07 August 2017].

Fama, E. & French, K. (2004), "The Capital Asset Pricing Model: Theory and Evidence", *The Journal of Economic Perspectives*, vol. 18, no. 3, pp. 25-46.

Filbeck, G., Holzhauer, H.M. & Zhao, X. (2014), "Using Social Responsibility Ratings to Outperform the Market: Evidence from Long-Only and Active-Extension Investment Strategies", *Journal of Investing*, vol. 23, no. 1, pp. 79-96.

Fisher, K. (1992), "Overripe", Forbes, vol. 149, no. 8, p. 138.

Gandhi, K. & Perumal, R. (2016), "Performance of Selected Bank Mutual Fund Schemes Impact in Investors' Decision Making", *International Journal of Advances Research in Management and Social Sciences*, vol. 5, no. 3, pp 361-370.

Gang, J. & Qian, Z. (2016), "Risk-Adjusted Performance of Mutual Funds: Evidence from China", *Emerging Market Finance & Trade*, vol. 52, no. 9, pp 2056-2068.

Garcia, A., Medes-Da-Silva, W. & Orsato, R. (2017), "Sensitive industries produce better ESG performance: Evidence from emerging markets", *Journal of Cleaner Production*, no. 150, pp 135-147.

Gasser, S., Rammerstorfer, M. & Weinmayer, K. (2017), "Markowitz revisited: Social portfolio engineering", *European Journal of Operational Research*, vol. 258, no. 3, pp 1181-1190.

Hay, J. (2015), "Green bonds: not exactly SRI, nor yet ESG", *Globalcapital*, p. 192.

Hodnett, K. & Hsieh, H. (2012), "The Value of Growth at A Reasonable Price: Evidence from The Taiwan Stock Exchange", *The International Business & Economics Research Journal*, vol. 11, no. 9, p. 1041.

Hua, T. (2011), "Search for ESG alpha continues", *Pensions & Investments*, vol. 39, no. 2, p. 12.

Junkus, J. & Berry, T.D. (2015), "Socially responsible investing: a review of the critical issues", *Managerial Finance*, vol. 41, no. 11, pp. 1176-1201.

Kempf, A, & Osthoff, P. (2007), "The Effect of Socially Responsible Investing on Portfolio Performance", *European Financial Management*, vol.13, no.5, pp. 908-922.

Kim, W., & Mulvey, J. (2009), "Evaluating Style Investment - Does a Fund Market Defined along Equity Styles Add Value?", *Quantitative Finance*, vol. 9, no. 6, pp. 637-651.

Kim, Y. & Statman, M. (2012), "Do Corporations Invest Enough in Environmental Responsibility?", *Journal of Business Ethics*, vol. 105, no. 1, pp. 115-129.

Kleiner, A. (1991), "What Does It Mean to Be Green?", *Harvard Business Review*, vol. 69, no. 4, pp. 38-47.

Kłobukowska, J. (2017), "Socially Responsible Investment in Asia", *Copernican Journal* of Finance and Accounting, vol. 6, no. 1, pp. 55-65.

Kohler, A. & Wittig, H. (2014), "Rethinking Portfolio Rebalancing: Introducing Risk Contribution Rebalancing as an Alternative Approach to Traditional Value-Based Rebalancing Strategies", *Journal of Portfolio Management*, vol. 40, no. 3, pp. 34-46.

Lai, T. & Stohs, M.H. (2015), "Yes, CAPM Is Dead", *International Journal of Business*, vol. 20, no. 2, pp. 144-158.

Lee, H., Cheng, F. & Chong, S. (2016), "Markowitz Portfolio Theory and Capital Asset Pricing Model for Kuala Lumpur Stock Exchange: A Case Revisited", *International Journal of Economics and Financial Issues*, vol. 6, no. 3.

Lee, S. (2012), "The performance gap in UK property returns", *Journal of European Real Estate Research*, vol. 5, no. 2, pp. 121-134.

Lesser, K., Lobe, S. & Walkshäusl, C. (2014), "Green and socially responsible investing in international markets", *Journal of Asset Management*, vol. 15, no. 5, pp. 317-331.

Liu, Z., & Wang, J. (2010a), "Time, risk, and investment styles", *Financial Services Review*, vol. 19, no. 4, pp. 323-336.

Liu, Z. & Wang, J. (2010b), "Value, Growth, and Style Rotation Strategies in the Long Run", *Journal of Financial Service Professionals*, vol. 64, no. 6, pp. 66-74.

Mallett, J.E. & Michelson, S. (2010), "Green Investing: Is it Different from Socially Responsible Investing?", International Journal of Business, vol. 15, no. 4, pp. 395-410.

Markowitz, H. (1952). "Portfolio selection", *The Journal of Finance*, vol. 7, no.1, pp. 77–91.

Marlowe, J. (2014), "Socially Responsible Investing and Public Pension Fund Performance", *Public Performance & Management Review*, vol. 38, no. 2, pp. 337-358.

Moerloose, S. & Giot, P. (2011), "Style investing and momentum investing: A case study", *Journal of Asset Management*, vol. 12, no. 6, pp. 407-417.

Muñoz, F., Vargas, M. & Marco, I. (2014), "Environmental Mutual Funds: Financial Performance and Managerial Abilities", *Journal of Business Ethics*, vol. 124, no.4, pp 551-569.

Napach, B. (2016), "Investing in ESG, SRI or Impact Funds? Do You Know the Difference?", ThinkAdvisor.

Nezlobin, A., Rajan, M.V. & Reichelstein, S. (2016), "Structural properties of the priceto-earnings and price-to-book ratios", *Review of Accounting Studies*, vol. 21, no. 2, pp. 438-472.

Nofsinger, J. & Varma, A. (2014), "Socially responsible funds and market crises", *Journal of Banking & Finance*, vol. 48, pp. 180–93.

MSCI (2014), Intangible Value Assessment (IVA) Methodology [ONLINE] available at: https://www.msci.com/documents/10199/242721/IVA Methodology SUMMARY.pdf [Accessed 07 August 2017]. MSCI (2017a), ESG Ratings Methodology [ONLINE] available at: https://www.msci.com/documents/10199/123a2b2b-1395-4aa2-a121-ea14de6d708a [Accessed 07 August 2017].

MSCI (2017b), MSCI ACWI Index [ONLINE] available at: https://www.msci.com/documents/1296102/1362201/MIS-ACWI.pdf/1a1e5e53-3dd8-40f3-8def-544258408cae [Accessed 07 August 2017].

Pallant, J. (2016), SPSS survival manual: a step by step guide to data analysis using the SPSS program, 6<sup>th</sup> Ed., McGraw Hill, Maidenhead.

Panopoulou, E. & Plastira, S. (2014), "Fama French factors and US stock return predictability", *Journal of Asset Management*, vol. 15, no. 2, pp. 110-128.

Petajisto, A. (2013), "Active Share and Mutual Fund Performance", Financial Analysts Journal, vol. 69, no. 4, pp. 73-93.

Pratt, J. (2011), "Why the Fama-French Factors Work and Serial Correlation too", *Academy of Accounting and Financial Studies Journal*, vol. 15, no. 3, pp. 53-72.

Prudhomme, J. (2015), *SRI and ESG – is there a difference?* [ONLINE] Available at: <a href="http://www.engagedinvestor.co.uk/sri-and-esg-is-there-a-difference/14746599.article">http://www.engagedinvestor.co.uk/sri-and-esg-is-there-a-difference/14746599.article</a> [Accessed 07 August 2017].

Przychodzen, J., Gómez-Bezares, F., Przychodzen, W. & Larreina, M. (2016), "ESG Issues among Fund Managers—Factors and Motives", *Sustainability*, vol. 8, no. 10, pp. 1078-1097.

Puopolo, G.W., Teti, E. & Milani, V. (2015), "Does the market reward for going green?", *Journal of Management Development*, vol. 34, no. 6, pp. 729-742.

Quinlan, C. (2011), Business *Research Methods*, South-Western Cengage Learning, Andover.

Reveilli, C. & Viviani, J.L. (2015), "Financial performance of socially responsible investing (SRI): what have we learned? A meta-analysis", *Business Ethics: A European Review*, vol. 24, no. 2, pp. 158 -185.

Sabbaghi, O. (2011), "The behaviour of green exchange-traded funds", *Managerial Finance*, vol. 37, no. 5, pp. 426-441.

Saunders, M., Lewis, P., & Thornhill, A. (2015), *Research Methods for Business Students*, 7<sup>th</sup> Ed. Pearson, New York.

Schröder, M. (2007), "Is there a Difference? The Performance Characteristics of SRI Equity Indices", *Journal of Business Finance & Accounting*, vol.34, no. 1/2, pp. 331-348.

Semenova, N. & Hassel, L. (2015), "On the Validity of Environmental Performance Metrics", *Journal of Business Ethics*, vol.132, no.2, pp. 249-258.

Social Investment Forum (2016), 2016 Report on US Sustainable, Responsible and Impact Investing Trends, [Online] available at: <u>http://www.ussif.org/files/SIF\_Trends\_16\_Executive\_Summary(1).pdf</u> [Accessed 24 January 2017].

Statman, M. & Glushkov, D. (2016), "Classifying and Measuring the Performance of Socially Responsible Mutual Funds", *Journal of Portfolio Management*, vol. 42, no. 2, pp. 140-151.

Tarmuji, I., Maelah, R. & Tarmuji, N. (2016), "The Impact of Environmental, Social and Governance Practices (ESG) on Economic Performance: Evidence from ESG Score", *International Journal of Trade, Economics and Finance*, vol. 7, no. 3, pp. 67-74.

Vo, D. (2015), "Which Factors Are Priced? An Application of the Fama French Three-Factor Model in Australia", *Economic Papers*, vol. 34, no. 4, pp. 290-301.

Vyšniauskas, P. & Rutkauskas, A. (2014), "Performance Evaluation of Investment (Mutual) Funds", *Business: Theory & Practice. 2014*, vol.15, no4, pp. 398-407.

Zacks, L. (2014). What Investors Gain from Large, Mid & Small-Cap Growth Funds -Mutual Fund Commentary. [Online] available at: <u>http://www.nasdaq.com/article/what-investors-gain-from-large-mid-small-cap-</u> growth-funds-mutual-fund-commentary-cm382323 [Accessed 3rd August 2017].

Zaremba, A. (2016), "Is there a low-risk anomaly across countries?", *Eurasian Economic Review*, vol. 6, no. 1, pp. 45-65.

# Appendices

## Appendix A: Descriptive Statistics

 Table 9.1: GGL versus NGL 3-Year Descriptive Statistics

	Portfoli	оТуре		Statistic	Std. Error
PortfolioReturns	GGL	Mean		1.0727553	.55552601
		95% Confidence Interval	Lower Bound	0550225	
		for Mean	Upper Bound	2.2005330	
		5% Trimmed Mean		1.1426710	
		Median		1.1658200	
		Variance		11.110	
		Std. Deviation		3.33315604	
		Minimum		-8.06519	
		Maximum		8.35359	
		Range		16.41878	
		Interquartile Range		4.02747	
		Skewness		458	.393
		Kurtosis		.731	.768
	NGL	Mean		.8505314	.52799465
		95% Confidence Interval	Lower Bound	2213547	
		for Mean	Upper Bound	1.9224175	
		5% Trimmed Mean		.9316069	
		Median		.7536200	
		Variance		10.036	
		Std. Deviation		3.16796790	
		Minimum		-7.20636	
		Maximum		6.77526	
		Range		13.98162	
		Interquartile Range		3.42809	
		Skewness		505	.393
		Kurtosis		.194	.768

## Table 9.2: GGL versus NGL 5-Year Descriptive Statistics

	Portfoli	оТуре		Statistic	Std. Error
PortfolioReturns	GGL	Mean		1.1989175	.37388644
		95% Confidence Interval	Lower Bound	.4507725	
		for Mean	Upper Bound	1.9470625	
		5% Trimmed Mean		1.2662852	
		Median		1.4664400	
		Variance		8.387	
		Std. Deviation		2.89611193	
		Minimum		-8.06519	
		Maximum		8.35359	
		Range		16.41878	
		Interquartile Range		3.88407	
		Skewness		518	.309
		Kurtosis		1.064	.608
	NGL	Mean		.9375393	.36540996
		95% Confidence Interval	Lower Bound	.2063557	
		for Mean	Upper Bound	1.6687230	
		5% Trimmed Mean		1.0110980	
		Median		.7113900	
		Variance		8.011	
		Std. Deviation		2.83045341	
		Minimum		-7.20636	
		Maximum		6.77526	
		Range		13.98162	
		Interquartile Range		3.20294	
		Skewness		440	.309
		Kurtosis		.390	.608

## Table 9.3: GGL versus NGL 10-Year Descriptive Statistics

	Portfoli	оТуре		Statistic	Std. Error
PortfolioReturns	GGL	Mean		.7442895	.33596061
		95% Confidence Interval	Lower Bound	.0790539	
		for Mean	Upper Bound	1.4095251	
		5% Trimmed Mean		.8850286	
		Median		1.2707250	
		Variance		13.544	
		Std. Deviation		3.68026410	
		Minimum		-10.15129	
		Maximum		9.76388	
		Range		19.91517	
		Interquartile Range		4.41947	
		Skewness		650	.221
		Kurtosis		.922	.438
	NGL	Mean		.5968060	.35921787
		95% Confidence Interval	Lower Bound	1144813	
		for Mean	Upper Bound	1.3080933	
		5% Trimmed Mean		.7658527	
		Median		.7861200	
		Variance		15.484	
		Std. Deviation		3.93503458	
		Minimum		-12.79405	
		Maximum		9.20588	
		Range		21.99993	
		Interquartile Range		4.14728	
		Skewness		774	.221
		Kurtosis		1.119	.438

## Table 9.4: GGL versus NGL 2008-2009 Descriptive Statistics

				Otatiatia	Otd. Ennen
	PortfolioType			Statistic	Sta. Error
PortfolioReturns	GGL	Mean		5646646	1.17763011
		95% Confidence Interval	Lower Bound	-3.0007781	
		for Mean	Upper Bound	1.8714489	
		5% Trimmed Mean		5901525	
		Median		.2431450	
		Variance		33.284	
		Std. Deviation		5.76918576	
		Minimum		-10.15129	
		Maximum		9.76388	
		Range		19.91517	
		Interquartile Range		8.80801	
		Skewness		185	.472
		Kurtosis		-1.036	.918
	NGL	Mean		9378675	1.26978921
		95% Confidence Interval for Mean	Lower Bound	-3.5646266	
			Upper Bound	1.6888916	
		5% Trimmed Mean		8468981	
		Median		.1583400	
		Variance		38.697	
		Std. Deviation		6.22067129	
		Minimum		-12.79405	
		Maximum		9.20588	
		Range		21.99993	
		Interquartile Range		9.94018	
		Skewness		283	.472
		Kurtosis		928	.918

## Table 9.5: GVL versus NVL 3-Year Descriptive Statistics

	PortfolioType			Statistic	Std. Error
PortfolioReturns	GVL	Mean		1.0229503	.61903443
		95% Confidence Interval for Mean	Lower Bound	2337564	
			Upper Bound	2.2796570	
		5% Trimmed Mean		1.1731191	
		Median		1.6047050	
		Variance		13.795	
		Std. Deviation		3.71420658	
		Minimum		-7.92231	
		Maximum		7.88602	
		Range		15.80833	
		Interquartile Range		3.99520	
		Skewness		672	.393
		Kurtosis		.485	.768
	NVL	Mean		.9605736	.63167137
		95% Confidence Interval for Mean	Lower Bound	3217874	
			Upper Bound	2.2429347	
		5% Trimmed Mean		1.0463788	
		Median		1.4397150	
		Variance		14.364	
		Std. Deviation		3.79002822	
		Minimum		-8.67651	
		Maximum		9.17196	
		Range		17.84847	
		Interquartile Range		3.92853	
		Skewness		530	.393
		Kurtosis		.529	.768

## Table 9.6: GVL versus NVL 5-Year Descriptive Statistics

PortfolioType				Statistic	Std. Error
PortfolioReturns	GVL	Mean		1.2758543	.42974464
		95% Confidence Interval for Mean	Lower Bound	.4159373	
			Upper Bound	2.1357714	
		5% Trimmed Mean		1.4167026	
		Median		1.8735300	
		Variance		11.081	
		Std. Deviation		3.32878767	
		Minimum		-7.92231	
		Maximum		7.88602	
		Range		15.80833	
		Interquartile Range		3.64231	
		Skewness		790	.309
		Kurtosis		.822	.608
	NVL	Mean		1.0250018	.43409257
		95% Confidence Interval for Mean	Lower Bound	.1563846	
			Upper Bound	1.8936191	
		5% Trimmed Mean		1.0911143	
		Median		1.4114650	
		Variance		11.306	
		Std. Deviation		3.36246662	
		Minimum		-8.67651	
		Maximum		9.17196	
		Range		17.84847	
		Interquartile Range		3.70455	
		Skewness		441	.309
		Kurtosis		.743	.608

## Table 9.7: GVL versus NVL 10-Year Descriptive Statistics

	PortfolioType			Statistic	Std. Error
PortfolioReturns	GVL	Mean		.6417330	.41060813
		95% Confidence Interval for Mean	Lower Bound	1713121	
			Upper Bound	1.4547781	
		5% Trimmed Mean		.6787404	
		Median		.9837250	
		Variance		20.232	
		Std. Deviation		4.49798669	
		Minimum		-11.03146	
		Maximum		21.48895	
		Range		32.52041	
		Interquartile Range		5.29265	
		Skewness		.266	.221
		Kurtosis		3.485	.438
	NVL	Mean		.4892132	.40210515
		95% Confidence Interval for Mean	Lower Bound	3069952	
			Upper Bound	1.2854215	
		5% Trimmed Mean		.5590764	
		Median		1.1029150	
		Variance		19.403	
		Std. Deviation		4.40484126	
		Minimum		-15.26974	
		Maximum		15.54707	
		Range		30.81681	
		Interquartile Range		4.73085	
		Skewness		278	.221
		Kurtosis		1.687	.438

Descriptives

## Table 9.8: GVL versus NVL 2008-2009 Descriptive Statistics

	Portfoli	оТуре	Statistic	Std. Error	
PortfolioReturns	GVL	Mean		3078013	1.52570019
		95% Confidence Interval for Mean	Lower Bound	-3.4639526	
			Upper Bound	2.8483501	
		5% Trimmed Mean		8183965	
		Median		1985900	
		Variance		55.866	
		Std. Deviation		7.47437392	
		Minimum		-11.03146	
		Maximum		21.48895	
		Range		32.52041	
		Interquartile Range		10.89737	
		Skewness		.901	.472
		Kurtosis		1.680	.918
	NVL	Mean		6191779	1.44464769
		95% Confidence Interval for Mean	Lower Bound	-3.6076593	
			Upper Bound	2.3693035	
		5% Trimmed Mean		6952806	
		Median		-1.5725750	
		Variance		50.088	
		Std. Deviation		7.07729938	
		Minimum		-15.26974	
		Maximum		15.54707	
		Range		30.81681	
		Interquartile Range		10.56976	
		Skewness		.192	.472
		Kurtosis		.087	.918

## Table 9.9: GGM versus NGM 3-Year Descriptive Statistics

	Statistic	Std. Error			
PortfolioReturns	GGM	Mean		.9787814	.59263567
		95% Confidence Interval for Mean	Lower Bound	2243330	
			Upper Bound	2.1818958	
		5% Trimmed Mean		1.0539112	
		Median		1.2526800	
		Variance		12.644	
		Std. Deviation		3.55581404	
		Minimum		-8.41112	
		Maximum		8.72171	
		Range		17.13283	
		Interquartile Range		4.04090	
		Skewness		354	.393
		Kurtosis		.705	.768
	NGM	Mean		.8867864	.54677145
		95% Confidence Interval	Lower Bound	2232187	
			Upper Bound	1.9967914	
		5% Trimmed Mean		.9950607	
		Median		.9361200	
		Variance		10.763	
		Std. Deviation		3.28062867	
		Minimum		-7.93010	
		Maximum		7.16734	
		Range		15.09744	
		Interquartile Range		3.39400	
		Skewness		680	.393
		Kurtosis		.784	.768

### Descriptives
# Table 9.10: GGM versus NGM 5-Year Descriptive Statistics

	Portfoli	оТуре		Statistic	Std. Error
PortfolioReturns	GGM	Mean		1.1826483	.40536012
		95% Confidence Interval	Lower Bound	.3715246	
		for Mean	Upper Bound	1.9937721	
		5% Trimmed Mean		1.2322189	
		Median		1.4632700	
		Variance		9.859	
		Std. Deviation		3.13990600	
		Minimum		-8.41112	
		Maximum		8.72171	
		Range		17.13283	
-		Interquartile Range		3.50555	
		Skewness		385	.309
		Kurtosis		1.054	.608
	NGM	Mean		.9211603	.36999569
		95% Confidence Interval for Mean	Lower Bound	.1808007	
			Upper Bound	1.6615200	
		5% Trimmed Mean		1.0087865	
		Median		.9914000	
		Variance		8.214	
		Std. Deviation		2.86597430	
		Minimum		-7.93010	
		Maximum		7.16734	
		Range		15.09744	
		Interquartile Range		2.96504	
		Skewness		601	.309
		Kurtosis		1.149	.608

# Table 9.11: GGM versus NGM 10-Year Descriptive Statistics

	PortfolioType			Statistic	Std. Error
PortfolioReturns	GGM	Mean		.7265368	.40757655
		95% Confidence Interval	Lower Bound	0805055	
		for Mean	Upper Bound	1.5335791	
		5% Trimmed Mean		.8746099	
		Median		1.4246500	
		Variance		19.934	
		Std. Deviation		4.46477745	
		Minimum		-18.06888	
		Maximum		15.43051	
		Range		33.49939	
		Interquartile Range		4.80244	
		Skewness		720	.221
		Kurtosis		3.066	.438
	NGM	Mean		.6942598	.39393029
		95% Confidence Interval for Mean	Lower Bound	0857615	
			Upper Bound	1.4742812	
		5% Trimmed Mean		.8452907	
		Median		.9949750	
		Variance		18.622	
		Std. Deviation		4.31529012	
		Minimum		-16.80142	
		Maximum		15.21151	
		Range		32.01293	
		Interquartile Range		4.27128	
		Skewness		681	.221
		Kurtosis		2.910	.438

# Table 9.12: GGM versus NGM 2008-2009 Descriptive Statistics

	Portfoli	оТуре	Statistic	Std. Error	
PortfolioReturns	GGM	Mean		2499742	1.56644106
		95% Confidence Interval	Lower Bound	-3.4904044	
		for Mean	Upper Bound	2.9904560	
		5% Trimmed Mean		1364688	
		Median		1.7635550	
		Variance		58.890	
		Std. Deviation		7.67396261	
		Minimum		-18.06888	
		Maximum		15.43051	
		Range		33.49939	
		Interquartile Range		10.34966	
		Skewness		345	.472
		Kurtosis		.175	.918
	NGM	Mean		3374079	1.51926851
		95% Confidence Interval for Mean	Lower Bound	-3.4802543	
			Upper Bound	2.8054384	
		5% Trimmed Mean		2752179	
		Median		1.3435600	
		Variance		55.396	
		Std. Deviation		7.44286524	
		Minimum		-16.80142	
		Maximum		15.21151	
		Range		32.01293	
		Interquartile Range		10.53959	
		Skewness		256	.472
		Kurtosis		.017	.918

# Table 9.13: GVM versus NVM 3-Year Descriptive Statistics

		-			
	Portfoli	PortfolioType			Std. Error
PortfolioReturns	G∨M	Mean		.9795714	.61276131
		95% Confidence Interval	Lower Bound	2644002	
		for Mean	Upper Bound	2.2235430	
		5% Trimmed Mean		1.0297667	
		Median		1.1489250	
		Variance		13.517	
		Std. Deviation		3.67656786	
		Minimum		-8.79546	
		Maximum		10.05055	
		Range		18.84601	
		Interquartile Range		3.68386	
		Skewness		145	.393
		Kurtosis		1.194	.768
	NVM	Mean		1.0009653	.71296464
		95% Confidence Interval for Mean	Lower Bound	4464299	
			Upper Bound	2.4483604	
		5% Trimmed Mean		1.0649457	
		Median		1.0381400	
		Variance		18.299	
		Std. Deviation		4.27778782	
		Minimum		-8.85773	
		Maximum		9.45151	
		Range		18.30924	
		Interquartile Range		6.54371	
		Skewness		385	.393
		Kurtosis		218	.768

# Table 9.14: GVM versus NVM 5-Year Descriptive Statistics

	PortfolioType			Statistic	Std. Error
PortfolioReturns	GVM	Mean		1.1181453	.44426685
		95% Confidence Interval	Lower Bound	.2291694	
		for Mean	Upper Bound	2.0071213	
		5% Trimmed Mean		1.1951085	
		Median		1.3175900	
		Variance		11.842	
		Std. Deviation		3.44127625	
		Minimum		-8.79546	
		Maximum		10.05055	
		Range		18.84601	
		Interquartile Range		3.15511	
		Skewness		301	.309
		Kurtosis		1.193	.608
	NVM	Mean		.9956633	.48588569
		95% Confidence Interval for Mean	Lower Bound	.0234083	
			Upper Bound	1.9679184	
		5% Trimmed Mean		1.0467309	
		Median		.9031700	
		Variance		14.165	
		Std. Deviation		3.76365440	
		Minimum		-8.85773	
		Maximum		9.45151	
		Range		18.30924	
		Interquartile Range		4.01786	
		Skewness		334	.309
		Kurtosis		.080	.608

# Table 9.15: GVM versus NVM 10-Year Descriptive Statistics

	Portfoli	oType		Statistic	Std. Error
PortfolioReturns	GVM	Mean		6597898	45635889
		95% Confidence Interval	Lower Bound	- 2438464	
		for Mean	Upper Bound	1 5634259	
		5% Trimmed Mean		.5961304	
		Median		8618750	
		Variance		24 992	
		Std. Deviation		4 99916115	
		Minimum		-12 90925	
		Maximum		26 36477	
		Range		39.27402	
		Interguartile Range		4 57502	
		Skewness		4.57502	221
		Kurtosis		5.632	.221
	NVM	Mean		9492010	.450
		95% Confidence Interval for Mean	Lower Bound	.0402010	.40343790
			Lower Bound	0694525	
		50/ Trimmed Mean	Opper Bound	1./658545	
		5% Inmmed Mean		.8209947	
		Median		.8293600	
		Variance		25.773	
		Std. Deviation		5.07670870	
		Minimum		-14.19001	
		Maximum		25.31976	
		Range		39.50977	
		Interquartile Range		5.58385	
		Skewness		.592	.221
		Kurtosis		4.270	.438

# Table 9.16: GVM versus NVM 2008-2009 Descriptive Statistics

	Portfoli	іоТуре		Statistic	Std. Error
PortfolioReturns	GVM	Mean		.2913700	1.77894232
		95% Confidence Interval	Lower Bound	-3.3886526	
		for Mean	Upper Bound	3.9713926	
		5% Trimmed Mean		2895693	
		Median		-1.0199150	
		Variance		75.951	
		Std. Deviation		8.71500193	
		Minimum		-12.90925	
		Maximum		26.36477	
		Range		39.27402	
		Interquartile Range		9.52840	
		Skewness		1.023	.472
		Kurtosis		2.196	.918
	N∨M	Mean		.8666425	1.78764652
		95% Confidence Interval for Mean	Lower Bound	-2.8313861	
			Upper Bound	4.5646711	
		5% Trimmed Mean		.4374880	
		Median		7755300	
		Variance		76.696	
		Std. Deviation		8.75764365	
		Minimum		-14.19001	
		Maximum		25.31976	
		Range		39.50977	
		Interquartile Range		10.56397	
		Skewness		.711	.472
		Kurtosis		1.371	.918

Descriptives

#### Appendix B: Letter of Permission from MSCI



Standalone-ESGA\_00229961.0

August 9, 2017

Fergal Twomey MBA Student National College of Ireland | Mayor Street, IFSC Dublin 1, DO1 Y3000 Ireland

#### Re: Use of Certain MSCI proprietary information in an academic report

Dear Fergal

 We are in receipt of your request to utilize and reproduce certain proprietary data of MSCI ESG Research LLC ("<u>MSCI</u>") in a MBA Dissertation for the National College of Ireland tentatively titled: "The performance of green stocks versus non-green stocks" (the "<u>Report</u>").

 The MSCI data to be made available to you shall consist of MSCI ESG Ratings data (the "<u>Data</u>"). The Data will be made available by MSCI for the Report through <u>datafred</u>.

3. MSCI hereby grants you a non-exclusive, non-transferable, non-sublicensable, limited license to reproduce limited, non-substantial portions of the Data in the Report, subject to MSCI's right of review set forth in Paragraph 6 below. The Data is being provided to you solely for non-commercial, non-profit research purposes related to the Report and may not be used for any other purpose including any other report. You shall not use the Data or any part thereof to develop or derive any other data product or data service for distribution or sale. Any commercial use of the Data will require the prior written approval of MSCI, which approval may be withheld by MSCI in its sole discretion. For the avoidance of doubt, under no circumstances may you incorporate the entirety of the Data and/or a substantial portion of the Data in any of the Report.

 For any reproduction or use of the Data, you agree to place the following notice in close proximity to the Data contained in the Report:

> Reproduced by permission of MSCI ESG Research LLC ©2017 MSCI ESG Research LLC. All rights reserved.

In addition, you will place the following notice in the Report:

Although National College of Ireland information providers, including without limitation, MSCI ESG Research LLC and its affiliates (the "ESG Parties"), obtain information from sources they consider reliable, none of the ESG Parties warrants or guarantees the originality, accuracy and/or completeness of any data herein. None of the ESG Parties makes any express or implied warranties of any kind, and the ESG Parties hereby expressly disclaim all warranties of merchantability and fitness for a particular purpose, with respect to any data herein. None of the ESG Parties shall have any liability for any errors or omissions in connection with any data herein. Further, without limiting any of the foregoing, in no

> Ninth Floor • Ten Bishops Square • London, E1 6EG • United Kingdom T +44 20 7618 2000 • F +44 20 7618 2233

MSCI ESG Research (UK) Limited. Registered in England and Wales. Company Number: 08811287.



Standalone-ESGA 00229961.0

event shall any of the ESG Parties have any liability for any direct, indirect, special, punitive, consequential or any other damages (including lost profits) even if notified of the possibility of such damages. MSCI ESG Research LLC's data contained herein is used under license and may not be further used, distributed or disseminated without the express written consent of MSCI ESG Research LLC.

 You acknowledge that your receipt and use of the Data is subject to the following disclaimer and limitation of liability:

ALTHOUGH MSCI SHALL OBTAIN INFORMATION FOR INCLUSION IN THE DATA FROM SOURCES THAT MSCI CONSIDERS RELIABLE, THE DATA IS PROVIDED TO YOU "AS IS" AND NEITHER MSCI, ANY OF ITS AFFILIATES, ANY OF ITS INFORMATION PROVIDERS NOR ANY THIRD PARTY INVOLVED IN, OR RELATED TO, COMPILING, COMPUTING OR CREATING THE SERVICE (COLLECTIVELY, THE "MSCI PARTIES") MAKES ANY REPRESENTATION OR WARRANTY OF ANY KIND TO YOU OR ANY THIRD PARTY, EITHER EXPRESS OR IMPLIED, WITH RESPECT TO THE DATA, THE TIMELINESS THEREOF, THE RESULTS TO BE OBTAINED BY THE USE THEREOF OR ANY OTHER MATTER, FURTHER, THE MSCI PARTIES EXPRESSLY DISCLAIM, AND YOU WAIVE, ANY AND ALL IMPLIED WARRANTIES, INCLUDING, WITHOUT LIMITATION, WARRANTIES OF ORIGINALITY, ACCURACY, COMPLETENESS, TIMELINESS, NON-INFRINGEMENT, MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE.

YOU ASSUME THE ENTIRE RISK OF ANY USE YOU MAY MAKE, OR PERMIT OR CAUSE TO BE MADE, OF THE DATA AND ACKNOWLEDGE THAT DATA FIELDS MAY NOT BE CONSISTENT THROUGHOUT THE DATA. NO MSCI PARTY SHALL HAVE ANY LIABILITY TO YOU OR ANY THIRD PARTY FOR ANY DAMAGES OF ANY KIND ARISING OUT OF, OR IN CONNECTION WITH, THIS AGREEMENT OR THE SUBJECT MATTER HEREOF, ANY ERRORS, DELAYS, OMISSIONS OR INTERRUPTIONS OF OR RELATED TO THE DATA OR YOUR OR ANY THIRD PARTY'S USE OF OR INABILITY TO USE THE DATA OR ANY PORTION THEREOF, OR OTHERWISE ARISING OUT OF, OR IN CONNECTION WITH, THIS AGREEMENT, WHETHER DIRECT, INDIRECT, INCIDENTAL, SPECIAL, PUNITIVE OR CONSEQUENTIAL (INCLUDING, WITHOUT LIMITATION, LOSS OF USE, LOSS OF PROFITS OR REVENUES OR OTHER ECONOMIC LOSS OF LICENSEE OR ANY THIRD PARTY), WHETHER IN TORT (INCLUDING, WITHOUT LIMITATION, NEGLIGENCE OR STRICT LIABILITY), CONTRACT OR OTHERWISE, AND WHETHER OR NOT THE MSCI PARTY HAS BEEN ADVISED OF, OR OTHERWISE MIGHT HAVE ANTICIPATED, THE POSSIBILITY OF SUCH DAMAGES.

6. You agree to provide MSCI with a draft of the Report prior to publication so that MSCI may review and approve the amount and scope of your use and disclosure of any Data. You will also provide MSCI with a copy of the Report upon publication and hereby grant MSCI a fully paid-up, royalty-free, non-exclusive, perpetual, irrevocable, worldwide, non-transferable, non-sublicensable license to use and distribute those portions of the Report that contain or utilize any ESG Data or information.

7. You acknowledge that MSCI is the sole and exclusive owner (or licensee) of the Data and all trade secrets, copyrights, trademarks and other intellectual property rights in or to the Data. No right or license in or to the Data is granted except as expressly provided herein and you will not make any use of

> Ninth Floor • Ten Bishops Square • London, E1 6EG • United Kingdom T+44 20 7618 2000 • F +44 20 7618 2233

MSCI ESG Research (UK) Limited. Registered in England and Wales. Company Number: 08811287.



Standalone-ESGA\_00229961.0

the Data not expressly authorized herein. This permission is limited by its terms, including that the material only be reproduced and used in connection with the publication.

8. This agreement shall be governed exclusively by, and construed exclusively in accordance with, the laws of New York, without regard to its conflict or choice of laws principles. You irrevocably consent to the exclusive jurisdiction of courts of competent jurisdiction located in New York County, New York for the adjudication of all disputes hereunder, and you consent to personal jurisdiction in such courts.

Please indicate your acceptance of the terms and conditions set forth above by countersigning in the space indicated below and returning this letter agreement to MSCI, c/o <u>cedric.lagrange@msci.com</u> via electronic pdf copy.

Sincerely yours,

MSCI ESG Research LLC

AGREED: Fergal Twomey

Signature:\_\_\_\_\_

Name:	
Title:	

]

Date:\_\_\_\_\_

Ninth Floor - Ten Bishops Square - London, E1 6EG - United Kingdom T +44 20 7618 2000 - F +44 20 7618 2233